

STUDENT PERCEPTIONS OF THE PREDICTORS OF CUSTOMER PURCHASE INTENTION OF COUNTERFEIT PRODUCTS By TINASHE CHUCHU STUDENT NUMBER: 731094

A Masters dissertation submitted in partial fulfilment of the requirements for the degree of the Master of Commerce (Marketing) in The School of Economic and Business Sciences, The University of the Witwatersrand.

Supervisor: Professor Richard Chinomona

19 MAY 2015

DECLARATION

- The work in this dissertation is my own original work.
- All of the sources which were used or referred to have been documented and recognised.
- This dissertation has not been previously submitted in full or partial fulfilment of the requirements for an equivalent or higher qualification at any other recognised education institution.

Cel

Tinashe Chuchu 19 May 2015

ACKNOWLEGDMENTS

Many people helped me throughout my research to make it the best it could possibly be. I cannot mention all of them by name but would like to thank them here. I would wish in particular, to thank my family for supporting me and giving me the opportunity to undertake my postgraduate studies – without my family I would not have been able to complete this dissertation. I also wish to thank my friends, who helped me throughout my studies, as well as all the participants who took part in my research. Most importantly, special mention goes to Mrs. Irma Cheval, and my supervisor, Professor Richard Chinomona. Through his guidance, wisdom and expertise in the areas of research I covered, completing this Masters dissertation was made possible.

ABSTRACT

The research investigated the use of the two potential predictors of customer purchase intention of counterfeit products (perceived behavioural control and price-quality inference of counterfeit products) as a means of establishing whether these were related to customer attitudes towards economic benefits of purchasing counterfeit products, which ultimately lead to purchase intention of counterfeit products. The study reviewed prior literature on counterfeits purchasing. Probability sampling was used to select respondents, and a research model was developed to measure the constructs used for the study. Field study was conducted in Johannesburg at the University of the Witwatersrand, and research data was collected from 380 registered students of the University of the Witwatersrand, 18 years and older. Using the SPSS 22 and AMOS 22 software program, structural equation modeling (SEM) was performed to analyse the data set. The results revealed that price-quality inference of counterfeit products had a stronger relationship with customer attitudes towards economic benefits of purchasing counterfeit products, than to perceived behavioural control implying that the customers were more likely to purchase counterfeits because of the perceived price bargain and not because of the perceived ease of purchasing counterfeits. The contribution of this study was to enhance the comprehension of existing literature on the relationship between the predictor variables (perceived behavioural control and price-quality inference of counterfeit products) and the outcome variable (purchase intention of counterfeit products).

Key words: counterfeits, attitudes, price-quality, purchase intention

TABLE OF CONTENTS

CHAPTER	PAGE
CHAPTER 1: OVERVIEW OF THE STUDY	2
1.1 INTRODUCTION	2
1.2 RESEARCH BACKGROUND	3
1.3 PROBLEM STATEMENT	6
1.4 PURPOSE OF THE STUDY	8
1.5 RESEARCH OBJECTIVES	8
1.5.1 Theoretical Objectives	8
1.5.2 Empirical Objectives	8
1.6 RESEARCH QUESTIONS	9
1.7 RESEARCH LIMITATIONS	9
1.8 JUSTIFICATION AND CONTRIBUTION OF THE STUDY	10
1.10 ORGANISATION OF THE THESIS	12
1.11 SUMMARY OF CHAPTER 1	13
CHAPTER 2: LITERATURE REVIEW	15
2.1 INTRODUCTION	15
2.2 THEORETICAL GROUNDING	15
2.2.1 Theory of Reasoned Action	15
2.2.2 Theory of Planned Behaviour	17
2.3 EMPIRICAL REVIEW	

2.3.1 Perceived Behavioural Control
2.3.2 Price – Quality Inference of Counterfeit Products
2.3.3 Attitudes Towards Economic Benefits of Purchasing Counterfeit Products20
2.3.4 Purchase Intention of Counterfeit Products
2.3 SUMMARY OF CHAPTER 222
CHAPTER 3: RESEARCH MODEL AND HYPOTHESIS STATEMENT
3.1 INTRODUCTION
3.2 HYPOTHESIS DEVELOPMENT
3.2.1 Perceived Behavioural Control and Purchase Intention of Counterfeit Products24
3.2.2 Perceived Behavioural Control and Attitudes Towards Economic Benefits of
Purchasing Counterfeit Products
3.2.3 Price-quality Inference of Counterfeit Products and Attitudes Towards Economic
Benefits of Purchasing Counterfeit Products26
3.2.4 Price-quality Inference of Counterfeit Products and Purchase Intention of
Counterfeit Products
3.2.5 Attitudes Towards Economic Benefits of Purchasing Counterfeit Products and
Purchase Intention of Counterfeit Products
3.3. SUMMARY OF CHAPTER 3
CHAPTER 4: RESEARCH METHODOLOGY AND DESIGN
4.1 INTRODUCTION
4.1.2 Research Philosophy: Positivist Paradigm
4.1.3 Quantitative Research

4.1.4 Sampling Design	
4.1.5 Target Population	
4.1.6 Sample Frame	
4.1.7 Sample Size	
4.1.8 Sample Method	
4.2 QUESTIONNAIRE DESIGN (MEASUREMENT INSTRUMENT)	
4.3 DATA COLLECTION	
4.3.1 Data Collection Techniques	
4.4 DATA ANALYSIS	
4.4.1 Structural Equation Modeling (SEM)	
4.5 DATA ANALYSIS APPROACH	40
4.5.1 Data Processing and Analysis	41
4.1 Confirmatory factor analysis (CFA)	41
4.2 Path modeling (PM)	41
4.5.2 Reliability and Validity of Measurement Scales	42
4.5.3 Tests of Measures and Accuracy Analysis Statistics	43
4.6 RELIABILITY TESTS	43
4.6.1 Cronbach's Alpha Test	44
4.6.2 Composite Reliability	45
4.7 VALIDITY TESTS	45
4.7.1 Construct Validity	46

4.7.2 Convergent Validity	46
4.7.3 Discriminant Validity	46
4.8 Model Fit/Acceptability	47
4.8.1 Confirmatory Factor Analysis	47
4.8.2 Path Modeling	51
4.9 ETHICAL CONSIDERATIONS	51
4.10 SUMMARY OF CHAPTER 4	52
CHAPTER 5: DATA ANALYSIS AND DISCUSSION OF RESULTS	55
5.1 INTRODUCTION	55
5.2. DATA SCREENING	56
5.3. DATA ANALYTICAL PROCEDURES	56
5.4 DESCRIPTIVE STATISTICS	57
5.4.1 Sample Description	57
5.4.2 Questionnaire Results	62
5.5 RELIABILITY TESTS	64
5.5.1 Cronbach's Alpha Test	64
5.5.2 Composite Reliability (CR)	64
5.5.3 Average Value Extracted (AVE)	66
5.6 VALIDITY TESTS	70
5.6.1 Convergent Validity	70
5.6.2 Discriminant Validity	70

5.7 Confirmatory Factor Analysis (CFA) models72
5.8. CONCEPTUAL MODEL FIT ASSESSEMENTS
5.10 Model Fit Results75
5.11 Discussion of Hypothesis Testing Results75
5.12 CHAPTER 5 SUMMARY76
CHAPTER 6: FINDINDGS, CONCLUSIONS, IMPILICATIONS AND FUTURE
RESEARCH79
6.1 INTRODUCTION79
6.2 IMPLICATIONS79
6.3 FUTURE RESEARCH AND LIMITATIONS
6.4 OVERALL RESEARCH CONTRIBUTION81
6.6 SUMMARY OF CHAPTER 682
LIST OF REFERENCES

LIST OF FIGURES

Figure 1: Diagrammatic Representation of Chapter 11
Figure 2: Research Flow11
Figure 3: Diagrammatic Representation of Chapter 214
Figure 4: Diagrammatic Representation of Chapter 323
Figure 5: Research Model24
Figure 6: Diagrammatic Representation of Chapter 4
Figure 7: Diagrammatic Representation of Research Methodology53
Figure 8: Diagrammatic Representation of Data Analysis Approach53
Figure 9: Diagrammatic Representation of Chapter 5
Figure 10: Statistical Analysis Procedures
Figure 11: Pie Chart Illustrating the Distribution of Males and Females in the Sample60
Figure 12: Bar Graph illustrating Educational Levels of the Respondents60
Figure 13: Bar Graph Illustrating Amount of Money Spent by Respondents61
Figure 14: Bar Graph illustrating Purchase Frequency of Respondents
Figure 15: Confirmatory Factor Analysis (CFA) Model72
Figure 16: Structural Model74
Figure 17: Diagrammatic Representation of Chapter 678

LIST OF TABLES

Table 1: Seizures of Imported Counterfeit and Pirated Products from 20 source Economies5
Table 2: Top Intellectual Property Rights Commodities Seized in the USA in 2006
Table 3: Definition of Variables 21
Table 4: Summary of Hypotheses
Table 5: Profile of University of the Witwatersrand Students
Table 6: Adapted and Original Measurement Items 36
Table 7: Reliability Test Criteria, Description and Acceptable Level
Table 8: Model Fit Criteria and Acceptable Fit Level 48
Table 9: Sample Demographic Profile
Table 10: Accuracy Analysis Statistics
Table 11: Diagrammatic Representation of Composite Reliability Calculations 67
Table 12: Diagrammatic Representation of Average Variance Extracted Calculations68
Table 13: Correlations between Constructs 71
Table 14: Model Fit Results
Table 15: Results of Structural Equation Model Analysis



Figure 1: Diagrammatic Representation of Chapter 1

CHAPTER 1: OVERVIEW OF THE STUDY

1.1 INTRODUCTION

Counterfeiting which refers to the production and sale of imitation branded products that closely resemble the original, has emerged as a serious problem for global marketers (Penz, Schlegelmilch and Stöttinger, 2008). Due to the increased competition in the market, brand imitation or piracy is one of the main problems that brand leaders face (Samgir, Sharma, and Sharma, 2013). Counterfeiting is a concept that became prominent in the 1970s (Veloutsou and Bian, 2008). Counterfeits infringe upon the legal rights' of registered trademarks owners' intellectual property (Lambkin and Tyndall, 2009; Stravinskiene, Dovaliene and Ambrazeviciute, 2013; Liao and Hsieh, 2013; Hoecht and Trott, 2014). This study focuses on customers' purchase intention of counterfeit products. Purchase Intention is defined as the consumers' will for purchase (Inkon, 2013). The intention to purchase can be viewed as a link between the buying behaviour and attitudes in understanding the buying behaviour of consumers, thus purchase intention has been considered to have a deep connection with buying behaviour (Inkon, 2013). The intention to purchase counterfeit luxury brands is influenced by attitudes toward buying counterfeits (Yoo and Lee, 2009). Attitude towards counterfeits of luxury brands have been found to influence purchase intention (Phau, 2009). The combination of attitude toward behaviour, subjective norm, and perception of behavioural control, lead to the formation of behavioural intention (Ajzen, 2002). Perceived behavioural control refers to the perceived ease or difficulty of performing behaviour (Ajzen, 2002). It is useful to consider perceived behavioral control in relation to intention, since it affects purchase intentions (Ajzen, 2002). Price-quality inference is the universal assumption across product categories that the level of the price cue is related positively to the quality level of the product (Zhou, Su, and Bao, 2002). It is believed that consumers may assume a positive relationship between price and product quality and rely on this price-quality

relationship or price-quality schema as a shortcut to making purchase decisions (Zhou, Su and Bao, 2002). Consumers who purchase counterfeit products are opting to pay for visual attributes and functions, without paying for the quality associated with the genuine product (Phau, Tea and Lee, 2008). Consumers that prefer counterfeit products bearing famous brand names feel that such actions accrue a sense of prestige (Phau, Tea and Lee, 2008). Lysonski and Durvasula (2008) stated that it is of great concern that some consumers do not perceive counterfeit purchasing as harmful to a particular industry.

1.2 RESEARCH BACKGROUND

1.2.1 The Counterfeit Industry

Counterfeit products have become a chief concern in the luxury product market, due to the expensive and prestigious characteristics of luxury brands (Wang and Song, 2013). This study used the term that is found most commonly in the reviewed literature on counterfeiting or counterfeits (Stravinskiene, Dovaliene and Ambrazeviciute, 2013). Counterfeiting is most evident in China (Hung, 2003; Naim, 2005; Minagawa, Trott and Hoecht, 2007). In particular, counterfeiters attempt to infringe upon brands, and may generate distorted information complexities to mislead customers (Quin, 2014). According to Howie (2010), up to three million customers purchase counterfeit goods daily. Counterfeiting prevails across the globe, accounting for at least 10% of world trade or worth of about 500 billion U.S dollars (Heffes, 2008; Yoo and Lee, 2009). Wang and Song (2013) have stated that with a counterfeit global market, estimated at around US\$600 billion, counterfeit luxury products account for 10% of this total (World Customs Organisation, 2004). According to Penz and Stöttinger (2008) counterfeiting occurs in two different forms, as deceptive and nondeceptive counterfeiting. Deceptive counterfeiting occurs in all the circumstances where the customer is unaware of being misled into buying a counterfeit product (Stravinskiene, Dovaliene and Ambrazeviciute, 2013). On the other hand, non-deceptive counterfeiting

occurs when customers are aware that the product is not genuine, but still make a conscious decision to purchase it (Eisend and Schuchert- Güler, 2006; Stravinskiene, Dovaliene and Ambrazeviciute, 2013). In discussing counterfeit items, scientific literature has used terms like counterfeiting, forging, brand piracy, imitation of brands or logos, fakes and replicas (Stravinskiene, Dovaliene and Ambrazeviciute, 2013). In 1997, the counterfeit market was worth \$299 billion dollars globally, according to Chakraborty, Allerd, Sukhdial and Bristol (1997). Quin (2014) pointed out that The World Customs Organisation (WCO, 2004) estimated that traded world merchandise valued at €512 billion, may have been counterfeits (WCO, 2004). Industries have blamed counterfeits for their deleterious effect on business, and the ethical costs associated with counterfeiting (Gino et al., 2010). The International anti-Counterfeiting Coalition estimates that counterfeiting is responsible for \$200 billion a year in lost jobs, taxes and sales (Furnham and Valgeirsson, 2007). It is likely that the demand for counterfeit products is caused by two main motives, namely, their relative affordability, and perceived quality (Budiman and Wijaya, 2014). Demand for counterfeits has been explored to some extent in the broader marketing literature (Kwong et al., 2003; Wilcox et al., 2009: Quin, 2014), with price, attitudes toward big branded companies, and the need for status signalling being cited as main factors driving counterfeit demand.

Past research has revealed that consumer's ethical attitudes can affect the likelihood of purchasing counterfeit goods (Norum and Cuno, 2011). Counterfeiting is considered to be an economic problem of international importance, which calls for a variety of counter-measures (Eisend and Schuchert-Güler, 2006). Counterfeit purchases are one of the most serious problems challenging luxury brand marketers (Hilton, Choi, and Chen, 2004; Poddar, Foreman, Banerjee, and Ellen, 2012). According to the International Chamber of Commerce (2008), the luxury brand industry reports a deficit in excess of \$12 billion annually due to

counterfeits. Poddar et al. (2012) further state that counterfeiting has become a worldwide problem, valued at more than \$250 billion, and accounting for almost 2% of world trade in 2009. Trade in counterfeit products poses severe negative consequences for customers, illicit manufactures and brand owners alike (Staake, Thiesse and Fleisch, 2009). Norum and Cuno (2011) have noted the production, distribution and consumption of counterfeit goods to be a global trend, escalating at an alarming rate. The IACC (2008) estimates that about five to seven percent of world's trade is in illegitimate goods (Norum and Cuno, 2011). Under deceptive counterfeiting, the customer is unaware of the purchasing of an imitation, rather than the original product, and cannot be held accountable for the behaviour, whereas with non-deceptive counterfeiting, the customer intentionally purchases fake products (Penz and Stöttinger, 2008). The Organisation for Economic Co-Operation and Development (OECD) covered the top 20 source economies for seizures of imported counterfeit and pirated products, noting that Asia is arising as the largest source for counterfeit and pirated products, with China as the single largest source economy (OECD, 2007). Table 1 below shows the seizures of imported counterfeit and pirated products from 20 nations in various regions.

Region of top 20	Number of source	Seizures
source economies	economies (countries) in	(% of total)
	region	
Asia (excl. Middle East)	12	69.7
Middle East	2	4.1
Africa	2	8
Europe	2	7
North America	1	1.1
South America	1	0.8
Top sources	20	79.2

 Table 1: Seizures of Imported Counterfeit and Pirated Products from 20 source

 Economies

Adopted from OECD (2007)

Table 2 below shows the top intellectual property rights commodities seized in the UnitedStates of America (USA) in 2006.

Commodity	Domestic value (USD)	Percent of total *	
Footwear	\$ 63,445,619	41%	
Apparel	\$ 24,320,976	16%	
Handbags/wallets/backpacks	\$ 14,750,201	9%	
Computers/hardware	\$ 14,287,989	9%	
Consumer electronics	\$ 7,057,034	5%	
Media	\$ 6,965,156	4%	
Headwear	\$ 3,257,963	2%	
Healthcare	\$ 3,092,919	2%	
Watches/parts	\$2,832,364	2%	
Pharmaceuticals	\$2,298,694	1%	
All other commodities	\$13,060,321	8%	
Total FY 06 Domestic Value	\$155,369,236		
Number of seizures	\$14,675		

Table 2: Top Intellectual Property Rights Commodities Seized in the USA in 2006

Adopted from Chaudhry, Zimmerman, Peters and Cordell (2009)

1.3 PROBLEM STATEMENT

It is of pronounced concern that consumers generally do not perceive the purchase of counterfeit products as harmful consumer behaviour, but only perceive the social benefits brought about by non-genuine products (Riquelme, Abbas and Rios, 2012). Lysonski and Durvasula (2008) have stated that it is of great concern that certain consumers do not perceive counterfeit purchasing as harmful to a particular industry. Counterfeiting ignores the effort of genuine manufacturers, and realises all profits without making any investment (Liao and Hsieh, 2013). In order to provide a sense of the magnitude of the counterfeit market, the number of illegitimate items seized at the European Union borders between 1990 and 2004 rose from 10 million to 103 million (Eisend and Schuchert- Güler, 2006). Although researchers Penz, Schlegelmilch and Stöttinger (2008) conducted a study on the purchase of counterfeit products, they focused on factors that drove the demand for fake products, as well as attempted to redress the imbalance that emerged from counterfeit demand attracting far less attention as compared to supply side remedies. However, Penz, Schlegelmilch and Stöttinger (2008) did not examine the way in which perceived behavioural control and price quality inference would possibly predict customers' intention to purchase counterfeit products. This dissertation intends to explore perceived behavioural control and price-quality inference of counterfeit products as predictors of customer attitudes towards the purchase of counterfeit products. This dissertation also attempts to address the relationship between perceived behavioural control, price-quality inference of counterfeit products, customer attitudes towards economic benefits of purchasing counterfeit products, and purchase intention of counterfeit products. According to Yoo and Lee (2009), counterfeit markets may help to stimulate demand for a product, and may provide symbolic benefits, such as providing consumers with social status through ownership of a brand name at a lower quality and cost, but counterfeiting's adverse socio-economic effects hinder innovation, employment, and trade.

1.4 PURPOSE OF THE STUDY

The purpose of this study is to investigate student perceptions of perceived behavioural control and price-quality inference of counterfeit products, as predictors of attitudes towards economic benefits of purchasing counterfeit products and purchase intention of counterfeit products.

1.5 RESEARCH OBJECTIVES

The section explores the theoretical and empirical research objectives of this study.

1.5.1 Theoretical Objectives

The following theoretical objectives were developed:

- to review literature on perceived behavioural control of counterfeit product purchase;
- to review literature on price-quality inference of counterfeit products;
- to review literature on attitudes towards economic benefits of purchasing counterfeit products; and
- to review literature on purchase intention of counterfeit products.

1.5.2 Empirical Objectives

Given the purpose of the study, the following empirical objectives were developed:

- to investigate the relationship between perceived behavioural control and purchase intention of counterfeit products;
- to investigate the relationship between perceived behavioural control and customer attitudes towards economic benefits of counterfeit product purchase;
- to investigate the relationship between price-quality inference of counterfeit products and customer attitudes towards economic benefits of counterfeit products purchase;

- to investigate the relationship between price-quality inference of counterfeit products and purchase intention of counterfeit products;
- to investigate the relationship between customer attitudes towards economic benefits of purchasing counterfeit products and purchase intention of counterfeit products.

1.6 RESEARCH QUESTIONS

- 1. Does perceived behavioural control have a relationship with customer purchase intention of counterfeit products?
- 2. Does perceived behavioural control have a relationship with customer attitudes towards economic benefits of purchasing counterfeit products?
- 3. Does the price-quality inference of counterfeit products have a relationship with customer attitudes towards economic benefits of purchasing counterfeit products?
- 4. Does the price-quality inference of counterfeit products have a relationship with customer purchase intention of counterfeit products?
- 5. Do customer attitudes towards the economic benefits of purchasing counterfeit products have a relationship with purchase intention of counterfeit products?

1.7 RESEARCH LIMITATIONS

Although this thesis is set to provide a contribution to both theoretical development and providing empirical evidence on the research constructs, there were limitations in terms of the population sampled. Since these respondents sampled were all registered students it implied that they had access to financial resources, and that a bias regarding their own perceptions of counterfeit products could have been created, due to education. The students' views can furthermore be generalised within the greater Johannesburg area. Other limitations were financial and time constraints, which lead the use of only registered students from the

University of the Witwatersrand as the sample frame. However, future research would have more informed results if conducted at a larger scale.

1.8 JUSTIFICATION AND CONTRIBUTION OF THE STUDY

This study will make a contribution to existing literature on:

- the relationship between perceived behavioural control and purchase intention of counterfeit products;
- the relationship between perceived behavioural control and attitudes towards economic benefits of purchasing counterfeit products;
- the relationship between price-quality inference of counterfeit products and attitudes towards economic benefits of purchasing counterfeit products;
- the relationship between price-quality inference of counterfeit products and purchase intention of counterfeit products;
- The relationship between attitudes towards economic benefits of purchasing counterfeit products and purchase intention of counterfeit products.

1.9 OVERVIEW OF THE RESEARCH PROCESS

A scientific research approach was adopted for this thesis, which comprised of a well thought out logical, organised and robust sequence of identifying problems, collecting data, analysing the data, and drawing valid conclusions. Firstly, the impetus of the research and main purpose of the study are elucidated. Secondly, literature relevant to the problem of research interest is reviewed. Thirdly, drawing from the reviewed theoretical and empirical literatures, a conceptual framework was built, and research hypotheses developed. Fourth, the development of hypotheses was then followed by measure of methodology, and data was collected thereafter. Fifth, the conceptualised model and posited hypotheses were empirically tested, using data collected, and the results were then interpreted. Lastly, closing remarks were provided, theoretical and practical implications of the thesis drawn, and suggestions for future research directions indicated. Figure 2 below illustrates the six-stage research procedure followed by the author of this thesis:

Figure 2: Research Flow



1.10 ORGANISATION OF THE THESIS

This study is composed of the following six chapters:

- An overview of the study in Chapter 1 provides the introduction and explores the research background. It also provides the purpose of this research, research objectives and research questions. Moreover, it illustrates the flow of the research and the structure of the thesis.
- Literature review in Chapter 2 reviews both theories and empirical studies related to the essence of the thesis. In particular, on the theoretical aspect of this thesis, theories such as the theory of reasoned action and the theory of planned behaviour. On the empirical aspect of this thesis, empirical studies regarding research constructs, i.e. perceived behavioural control, price-quality inference of counterfeit products, attitudes towards economic benefits of purchasing counterfeit products, and purchase intention of counterfeit products.
- Conceptual model and hypothesis development in Chapter 3, where, based on the research constructs grounded in theory, a conceptual model is built and research hypotheses developed.
- Research methodology, data analysis and results in Chapter 4 provides research sampling, measures and data collection procedure.
- Data measurement results in Chapter 5 details statistical techniques used to check for the measurement reliability, validity and the research model fit (confirmatory factor analysis - CFA) using SPSS 22 and AMOS 22 software. Hypotheses testing, where structural equation modeling (SEM), is used to empirically test the proposed research hypotheses. Finally, research findings are interpreted and presented.
- Conclusions, implications and overall thesis contributions are covered in Chapter 6: The overall concluding remarks informed by the findings of this thesis are thereafter

provided. Practical recommendations to the practice of distribution channel management are provided. Finally, future research directions are indicated, and the overall thesis contributions provided.

1.11 SUMMARY OF CHAPTER 1

Chapter 1 comprises ten sections. The first section begins with an introduction to the research, which focuses on the purchase intention of counterfeits, and thereafter explores the various definitions of counterfeit products. The second section explores the research background and takes a closer look at the counterfeit industry. The third section discusses the problem statement as the main motivation for this research. The forth section explores the purpose of the study. The fifth section outlines the research questions. In the sixth section, the limitations and challenges of the research are explored. The seventh section is made up of the justification of the study, as well as its intended contributions to the broader literature. The eighth section explores the overview of research, or the flow of research, that was used for this thesis. The ninth section details the organizational layout of the dissertation, and finally the tenth section of Chapter 1 concludes the chapter, by providing a summary.





CHAPTER 2: LITERATURE REVIEW

2.1 INTRODUCTION

Baker (2000) considers reviewing the current literature relevant to a research topic to be an essential first step and foundation for undertaking the research project. The main aim of the literature review was to discuss key issues that relate to current literature relevant to theoretical grounding and empirical literature of this dissertation. This chapter provides a detailed analysis of the relevant empirical studies and theoretical grounding that have dealt with the research topic, namely, student perceptions of the predictors of customer purchase intention of counterfeit products. This chapter discusses in detail the theories that were consulted for this study, namely the theory of reasoned action, and the theory of planned behaviour. The empirical literature review is also discussed in this chapter, whereby all the research constructs that were utilised for this study are explored, perceived behavioural control, price-quality inference of counterfeit products, attitudes towards economic benefits of purchasing counterfeit products, and purchase intention of counterfeit products.

2.2 THEORETICAL GROUNDING

For the purpose of this study, the following theoretical framework is employed.

2.2.1 Theory of Reasoned Action

The Theory of reasoned action (TRA) provides an essential fundamental conceptual model for probing human behaviour (Mishra, Akman and Mishra, 2014). The TRA is a welldeveloped and validated intention model that has been established successfully in predicting and elucidating behaviour (Liao, Lin and Liu, 2009). According to TRA, an individual's behaviour is driven by the intention to perform a given behaviour; intention, in turn, is jointly determined by the individual's attitude toward the behaviour (Liao, Lin and Liu, 2009). Thus, individuals who are highly motivated to engage in certain behaviour, are more likely to succeed at carrying it out (Christine, Henle, Reeve and Pitts, 2009). The purpose of the TRA is to investigate the relationship between attitude and behaviour (Mishra, Akman and Mishra, 2014). The Theory of Reasoned Action is arguably the most successful and well-validated psychological model for decision making for voluntary action to have been based on behavioural analysis (Houran, Lange and Lange, 2013). Phau, Sequeira and Dix (2009) have stated that the theory of reasoned action (TRA), in terms of counterfeit products purchase, would mean that the individual's decision to purchase counterfeit products is determined by that individual's intention to perform the act directly. The TRA posits the most immediate precursor of behaviour to be an individual's intention to behave, which is, in turn, determined by attitude toward target behaviour and subjective norms (Aleassa, Pearson and McClurg, 2011).

According to Hansen, Jensen, and Solgaard (2004), the theory of reasoned action considers a customer's behaviour as determined by the customer's behavioural intention, whereas behavioural intention of the customer is a role of attitude toward the customer's behaviour. Kim, Jeong and Hwang (2012) have stated that the theory of reasoned action, the theory of planned behaviour, and the integrative model of behavioural prediction, predict that behaviour is determined by behavioural intentions, which in turn are influenced by several psychological factors. According to Guo and Feng (2012), the theory of reasoned action postulates a model conceptually applicable to the totality of human activities. The theory of reasoned action specifies what consumers perceive, and how perceptions guide what consumers do, and that this is reminiscent of earlier studies on the drive toward conformity (Guo and Feng, 2012).

2.2.2 Theory of Planned Behaviour

The Theory of Planned Behaviour (TPB) is a philosophy used to predict human behaviour (Prapavessis, Gaston & DeJesus, 2015). The theory of planned behaviour is an extension of TRA, which has drawn sustained interest in social psychology (Conner and Armitage, 1998). The TPB arose due to the limitations of theory of reasoned action in its model when dealing with behaviours over which people have incomplete voluntary control (Ajzen, 1991). Henle, Reeve and Pitts (2009) have pointed out that when individuals have a stronger motivation or intention to engage in a particular behaviour, the TPB proposes that individuals are more likely to follow through with that behaviour. The TPB is an extensively applied expectancyvalue model of attitude-behaviour relationships, which has seen a fair amount of success in predicting a variety of behaviours (Conner and Armitage, 1998), and has emerged as one of the most influential and prominent conceptual frameworks for the study of human action (Ajzen, 2002). Armitage and Conner (2001) have pointed out that the theory of planned behaviour (TPB) is essentially an extension of the theory of reasoned action (TRA), which includes measures of control belief and perceived behavioural control. It predicts the occurrence of a specific action, provided that the action is intentional (Francis, Eccles, Johnston, Walker, Grimshaw, Foy, Kaner, Smith, Bonetti, 2004). Cooke and Sheeran (2004) have stated that the theory of planned behaviour is perhaps the leading account of the relationship between cognitions and behaviour in social psychology. Francis et al. (2004) have pointed out that in order to predict whether an individual intends to do something, there is need to know whether that individual is in favour of performing that particular act, how much the individual feels the social pressure to do it, and whether the individual feels in control of the action in question. Ajzen (2002) has stated briefly, that according to the theory, human behaviour is guided by three kinds of considerations: beliefs about the likely consequences or other attributes of the behaviour (behavioural beliefs); beliefs about the

normative expectations of other people (normative beliefs); and beliefs about the presence of factors that may further or hinder performance of the behaviour (control beliefs).

2.3 EMPIRICAL REVIEW

The following empirical review of the research constructs was conducted.

2.3.1 Perceived Behavioural Control

Perceived behavioural control is the independently "perceived simplicity or difficulty of performing a specific behaviour" (Ajzen, 2002). Armitage and Conner (2001) have pointed out that Ajzen (1991) extended the TRA to include a measure of perceived behavioural control, a variable that had received a great deal of attention in social cognition models designed to predict health behaviours (e.g. health belief model, protection motivation theory). According to Ajzen (2001) perceived behavioural control has arisen as one of the most influential and prevalent conceptual frameworks for the study of human action (Ajzen, 2001). Perceived behavioural control (information, time, ability to solve problems) positively affects intention to purchase counterfeits (Penz and Stöttinger, 2005; Cheng, Fu and Tu, 2011). According to Cheng, Fu, and Tu (2011), as far as the purchase of counterfeits is concerned, the time required to access those counterfeits and the customer's ability to solve difficulties that may arise in product purchases, are both important factors.

2.3.2 Price – Quality Inference of Counterfeit Products

According to Huang, Lee, and Ho (2004), the two major differences consumers perceive between a counterfeit and genuine products, are the lower prices and the poorer guaranties, which are price and risk constructs that are likely to be important factors relating to attitude toward counterfeits. De Matos, Ituassu and Rossi (2007) have stated that price difference is an important variable when choosing a counterfeit. Inference of quality by price level is a common perception among customers, and an important factor in consumer behaviour (De Matos, Ituassu and Rossi, 2007). In this sense, consumers' tendency to believe that "high (low) price means high (low) quality", becomes even more prevalent when there is little information about the product quality or the consumer is unable to judge product quality (De Matos, Ituassu and Rossi, 2007). Price-quality inference is the universal assumption across product categories that the level of the price cue is related positively to the quality level of the product (Zhou, Su, Bao, 2002). It is believed that consumers may assume a positive relationship between price and product quality, and that they may rely on this price quality relationship or price-quality schema as a shortcut to making purchase decisions (Lichtenstein et al., 1993; Zhou, Su and Bao, 2002).

According to Huang, Lee, and Ho (2004), counterfeits are usually sold at lower prices than original products, hence the greater the price-quality relationship for the consumer, the lower his/her perception of quality for the counterfeits. Huang, Lee, and Ho, (2004) added to state that customers who more strongly believe in the price-quality inference of products would have a more negative attitude toward counterfeits. According to the theory of planned behaviour, attitudes toward a particular action positively affect behavioural intentions (Yoo and Lee, 2009). Phau, Tea and Lee (2009) have established through research that consumers will select a counterfeit over a genuine product when there is a price advantage. Phau, Tea and Lee (2009) have mentioned that if counterfeit products are comparable to the genuine in all aspects and yet is superior in price offered, then consumers will choose counterfeits, as they provide the advantage of the status and quality attributes of brand-name products; and on the other hand, where other consumers might feel that although counterfeits are inferior to the original, their superior prices more than compensate for the shortfall in quality and performance.

2.3.3 Attitudes Towards Economic Benefits of Purchasing Counterfeit Products

Several researchers have explored the attitudes towards luxury goods and their counterfeit copies (Michaelidou and Christodoulides, 2011; Turunen and Laaksonen, 2011; Doss and Robinson, 2013). Michaelidou and Christodoulides (2011) have specifically focused on investigating the impact of ethical obligation and price consciousness on attitude towards counterfeit products, whilst Turunen and Laaksonen (2011) examined luxury consumption through exploring attributes of counterfeit branded products. Doss and Robinson (2013) focused on enhancing the comprehension of female consumers' luxury brand perceptions and counterfeits of that brand. Schiffman, Kanuk, and Wisenblit (2010) have defined attitude as a learned predisposition to behave in a consistently favourable or unfavourable manner with respect to a given object. Having an understanding about attitude is important, as it directly influences consumer behavior (Chaudary, Ahmed, Gill and Rizwan, 2014). According to Cheng, Fu and Tu (2011), attitudes towards purchasing counterfeits and perceived behavioural control both have a positive influence on the intention to purchase counterfeits and ultimately lead to behaviour.

Triandewil and Tjiptono (2013) have meanwhile pointed out that according to TRA and its extension, TPB, there is a connection between attitude, behavioural intention, and behaviour (Ajzen, 2005). Attitudes toward counterfeiting have been variously studied in the literature (Ang et al., 2001; Kwong et al., 2003; Wang et al., 2005) and are explained by TRA and TPB (Teah and Phau, 2008). In this regard, it is widely believed that human behaviour is determined by both social influences and dispositional (attitudinal) factors (Deci and Ryan, 2002; Aleassa, Pearson and McClurg, 2011). Yoo and Lee (2009) have stated that attitudes refer to the extent to which an individual has a favourable judgment of the behaviour in question, and are a direct indicator by which one's intention of conducting the specific behaviour can be predicted. Positive attitudes toward buying counterfeit products are

anticipated to affect purchase intention of genuine products negatively, and the purchase intention of counterfeit products positively (Yoo and Lee, 2009).

2.3.4 Purchase Intention of Counterfeit Products

According to the TRA, an individual's behavioral intention depends on his or her attitude toward the behaviour and the subjective norms associated with the behaviour (Zhan and He, 2012). The intention to buy counterfeit products is the tendency of individual willingness to buy and use the counterfeit products (Budiman and Wijaya, 2014). According to a study by Tan (2002), in which 377 Chinese software users where interviewed, it was established that purchase intentions are influenced by; the perceived moral intensity; degree of consequences; temporal use and social risk; the perceived commercial performance; and prosecution and moral judgment. Price is one of the most prevalent determinates of customer intention to purchase counterfeits over original brands (Triandewil and Tjiptono, 2013). Table 3 below shows the definitions of all the variables of the research model.

Construct	Definition	Sources	Item
Perceived behavioural	Perceived simplicity or difficulty of performing a specific	Ajzen (2002)	PBC 1-
control (PBC)	behaviour.		PBC 9
Attitudes towards	Consumer's psychological tendency to behave in a	Mathiesson et	ATT 1-
economic benefits of	consistently favourable or unfavourable fashion with	al. (2001)	ATT 6
purchasing counterfeit	regards to counterfeit products.		
products (ATT)			
Price-quality inference	The perceived price-quality relationship of a product.	Cronley and	PQ 1- PQ 3
of counterfeit products		Posava (2005)	
(PQ)			

 Table 3: Definition of Variables

The desire for customers to purchase a grey market	Bai et al.	PI 1- PI 4
product.	(2008)	
	The desire for customers to purchase a grey market product.	The desire for customers to purchase a grey market Bai et al. product. (2008)

2.3 SUMMARY OF CHAPTER 2

Chapter 2 was divided into three main sections. The first section discussed the theoretical grounding of the study that comprised of two theories; the Theory of Reasoned Action (TBA) and the Theory of Planned Behaviour (TPB). The second main section explored the study's empirical literature review of the research constructs in great detail. The research constructs included perceived behavioural control, price-quality inference of counterfeit products, attitudes towards economic benefits of purchasing counterfeit products, and finally, purchase intention of counterfeit products. The relationships and links among all the above mentioned constructs were also explored in great detail.



Figure 4: Diagrammatic Representation of Chapter 3

CHAPTER 3: RESEARCH MODEL AND HYPOTHESIS STATEMENT

3.1 INTRODUCTION

Drawing from the literature review explored in Chapter 2, in particular the theoretical and empirical literature, a research model was conceptualised. Hypothesised relationships between research constructs were developed for further empirical examination thereafter. In the conceptualised research model, perceived behavioural control (PBC) and price-quality inference of counterfeit products (PQ) are the predictor variables, attitudes towards economic benefits of purchasing counterfeit products (ATT) is the mediator variable, and purchase intention of counterfeit products (PI), the outcome variable. Figure 5 below illustrates the proposed conceptual model.





3.2 HYPOTHESIS DEVELOPMENT

Based on the above conceptual model, the following hypotheses are stated in the table below.

3.2.1 Perceived Behavioural Control and Purchase Intention of Counterfeit Products

According to (Ajzen, 1991) perceived behavioural control refers to people's perception of the ease or difficulty of performing the action in question. It is useful to consider perceived behavioral control in relation to intention, since it affects purchase intentions (Ajzen, 2002).

Perceived behavioural control proposes that the extent to which intentions lead to actual behaviour, depends partially on the amount of control of that individual has over that particular behaviour (Eagly and Chaiken, 1993). Penz and Stöttinger (2005) have since suggested that the higher the perceived behavioural control, the stronger the intention to purchase counterfeits. Conner and Armitage (1998) have however noted that connection between perceived behavioural control (PBC) and behaviour is complex, suggesting that a person is more likely to engage in (attractive/desirable) behaviours that he or she has control over, and is moreover prevented from carrying out behaviours that he or she does not have control over. Conner and Armitage (1998) have further stated that the more perceived behavioural control increases, the more the behaviour is likely to be performed, given that intention is held constant. Perceived behavioural control (information, time, ability to solve problems) positively affect intention to purchase counterfeits (Penz and Stöttinger, 2005; Cheng, Fu and Tu, 2011). Therefore, inferring from the literature and the empirical evidence abovementioned, the study hypothesises that:

H1: Perceived behavioural control has a positive relationship with purchase intention of counterfeit products

3.2.2 Perceived Behavioural Control and Attitudes Towards Economic Benefits of Purchasing Counterfeit Products

Conner and Armitage (1998) have defined attitudes as the complete evaluations of a given behaviour by the individual. According to Cheng, Fu and Tu (2011) attitudes towards purchasing counterfeits and perceived behavioural control both have a positive influence on the intention to purchase counterfeits, and to ultimately lead to a given behaviour. The combination of attitude toward behaviour, subjective norm and perception of behavioural control, lead to the formation of behavioural intention (Ajzen, 2002). According to Yoo and Lee's (2009) conceptual framework, attitudes toward buying counterfeits by economic
benefits positively influence customers' purchase intention of those counterfeits. Ajzen (1991) has stated that attitudes toward behaviour can be used to predict intentions to perform behaviours of different kinds, with a high degree of accuracy. Ajzen (1991) has continued further to state that subjective norms, perceived behavioural control, and intentions, together with perceptions of behavioural control, account for considerable variance in actual behaviour.

Therefore, inferring from the literature and the empirical evidence mentioned above, the study hypothesises that:

H2: Perceived behavioural control has a positive relationship with attitudes towards economic benefits of purchasing counterfeit products

3.2.3 Price-quality Inference of Counterfeit Products and Attitudes Towards Economic Benefits of Purchasing Counterfeit Products

Price-quality inference is the universal assumption across product categories that the level of the price cue is related positively to the quality level of the product (Zhou, Su, Bao, 2002).

It is believed that consumers may assume a positive relationship between price and product quality, and may rely on this price-quality relationship or price-quality schema as a shortcut to making purchasing decisions (Zhou, Su and Bao, 2002). The price-quality schema performs a critical role in consumer decision-making, where it affects the judgments of perceived quality, influences perceived value and purchase intentions, and determines information search and other aspects of consumer decision-making processes (Zhou, Su and Bao, 2002). Based on past studies on counterfeits, it has been established that consumers are more inclined towards choosing a counterfeit product over a genuine one, if there is a price advantage (Phau, Teah and Lee 2009). However other prior studies as that of (Grossman and Shapiro, 1988), imply that counterfeits can be categorised into two groups: one that will

consider a counterfeit over a genuine product, due to the price advantage given that all that aspects are similar; and the other group would consider the counterfeit product simply for the lower price compensating for the shortfall in quality. Therefore, inferring from the literature and the empirical evidence mentioned above, this study hypothesises that:

H3: price-quality inference of counterfeit products has a positive relationship with attitudes towards economic benefits of purchasing counterfeit products

3.2.4 Price-quality Inference of Counterfeit Products and Purchase Intention of

Counterfeit Products

According to Phau, Teah and Lee (2008), based on previous studies on counterfeit purchasing, consumers are more likely to purchase a counterfeit product over a genuine product if a price advantage exists. Grossman and Shapiro (1988) have meanwhile suggested that two types of counterfeit buyers exist, as far as price and quality inference is concerned. The first group perceives counterfeit products as being comparable to genuine products in all aspects, and yet is superior in price offered, then consumers will choose counterfeits, as they provide the status and quality attributes of brand-name products (Grossman and Shapiro, 1988). The second group perceive counterfeit products to be inferior to genuine products, where their superior prices more than compensate for the lack in quality and performance (Grossman and Shapiro, 1988). Therefore, inferring from the literature and the empirical evidence mentioned above, the study hypothesises that:

H4: Price-quality inference of counterfeit products has a positive relationship with purchase intention of counterfeit products

3.2.5 Attitudes Towards Economic Benefits of Purchasing Counterfeit Products and Purchase Intention of Counterfeit Products

According to Yoo and Lee (2009), attitudes are defined as to the extent to which one has a favourable judgement of the behaviour in question, and is an immediate indicator by which his or her intention of conducting the specific behaviour can be predicted. According to the assumptions of the theory of planned behaviour, attitudes toward an act positively affect behaviour (Yoo and Lee, 2009). Building on the theory of planned behaviour, Phau, Teah and Lee (2009) have suggested that attitudes determine purchase intention and purchase behaviour influences purchase intention. Yoo and Lee (2009) illustrated in their conceptual framework that attitudes towards the economic benefits of purchasing counterfeit products have an influence on purchase intention of counterfeit products. Cordell, Wongtada and Kieschnick (1996) have meanwhile stated that consumer involvement in counterfeit transactions facilitates illegal activity, and thus, an attitude-intention-behaviour linkage between the consumer's appreciation for lawfulness and willingness to buy counterfeits could explain consumer participation in the activity. Phau, Teah and Lee (2009) have pointed out that the accessibility of counterfeit products is necessary for purchase behaviour to occur, because the absence of access to counterfeit products would make their purchases difficult to perform. Therefore, inferring from the literature and the empirical evidence abovementioned, the study hypothesises that:

H5: Attitudes towards economic benefits of purchasing counterfeit products have a positive relationship with purchase intention of counterfeit products

Below Table 4 illustrates the hypotheses for the current study.

Table 4: Summary of Hypotheses

	SUMMARY OF HYPOTHESES	
H/No.	Hypothesis Statement	Hypothesized Relationship
	Perceived behavioural control has a positive relationship	
H1	with purchase intention of counterfeit products.	PBC→PI (+)
	Perceived behavioural control has a positive relationship	
H2	with attitudes towards economic benefits of purchasing	PBC → ATT (+)
	counterfeit products.	
	Price-quality inference of counterfeit products has a positive	
Н3	relationship with attitudes towards economic benefits of	PQ → ATT (+)
	purchasing counterfeit products.	
	Price-quality inference of counterfeit products has a positive	$PQ \rightarrow PI (+)$
H4	relationship with purchase intention of counterfeit products.	
	Attitudes towards economic benefits of purchasing	ATT→PI (+)
Н5	counterfeit products have a positive relationship with	
	purchase intention of counterfeit products.	

Key:

- PBC = Perceived behavioural control
- PQ = Price-quality inference of counterfeit products
- ATT = Attitudes towards economic benefits of purchasing counterfeit products
- PI = Purchase Intention of counterfeit products

3.3. SUMMARY OF CHAPTER 3

Chapter 3 provided the conceptualised model for the research study that constituted this thesis. Further to that, the research hypotheses for the research study were developed. The aim of this chapter was to illustrate the framework forming the basis of this study, as well as to formulate the proposed hypotheses, while supporting them with existing literature. The chapter comprised of two sections. In the first section, the research model was depicted and the hypothesised relationships indicated. In the second section, the hypothesised relationships were developed and further substantiated.



Figure 6: Diagrammatic Representation of Chapter 4

CHAPTER 4: RESEARCH METHODOLOGY AND DESIGN

4.1 INTRODUCTION

The previous chapter (Chapter 3) provided a detailed theoretical model that formed the basis of the research and explained the variables within the model that was adopted. Research methodology denotes the techniques used to acquire and analyse data to generate new knowledge (Petty, Thomson and Stew, 2012). Research methodologies have an influence on the validity and overview of a study, and play a crucial part in knowledge development (Yang, Wang and Su, 2006). A methodology maps out procedures for gathering information that will be used to answer questions and solve problems (Malhotra and Birks, 2007). The design of this study was quantitative in nature. Seven-point likert-scales were utilised in the collection of the data from respondents regarding their preferences and attitudes towards the purchase of counterfeit products. The researcher decided to use a seven-point likert-scale, because this study adopted questions from similar prior studies that had been measured by a seven point likert scale, where the researcher felt it was most appropriate to use the same measurement scales. A research questionnaire was self-administered to participants. The questionnaire was administered to students who had presented a University of the Witwatersrand student card to the researcher as proof of registration.

Understanding and use of relevant methodology for the study was essential in order to identify the unit of analysis, and to employ compatible methods that would provide the intended results. The research design explored:

- How data was to be acquired
- What instruments were employed
- How the instruments were utilised and
- The intended means for analysing data collected

4.1.2 Research Philosophy: Positivist Paradigm

According to Collins (2010), research philosophy refers to the advance and nature of knowledge. The concept of the paradigm is fundamental to the research process in all areas of study (Mangan, Lalwani and Gardner, 2004). The positivist paradigm is a philosophy in agreement with the empiricist view that knowledge sterns from human experience (Collins, 2010). Collins (2010) stated that positivism is an atomistic, ontological view (nature of reality) of the world comprising discrete, observable elements and events that interact in an observable, determined and regular manner.

4.1.3 Quantitative Research

A quantitative research approach was conducted for the purposes of this study. According to Polit and Hungler (1995), a quantitative approach utilises an organised procedure, along with techniques to gather information under controlled conditions, and highlights objectivity through statistical analysis. As such, a quantitative approach is more appropriate, given the nature of the current study.

4.1.4 Sampling Design

A sampling design should be simple to implement, efficient, and should cover various approaches to measure the sample to be generally applicable (Grafstrom, 2010).

4.1.5 Target Population

The identification of the study population is necessary for the formulation and running of any test (Klein and Meyskens, 2001). When defining a target population, a researcher should indicate clearly the characteristics of the target population that apply directly to the study. Students of the University of the Witwatersrand were used as the target population. Table 5 on the next page illustrates statistics on the characteristics of students of the University of the Witwatersrand, based on a survey conducted in 2010.

Total student enrolments	30833
Postgraduate students	31%
International students	10%
Male students	69%
Female students	31%

Table 5: Profile of University of the Witwatersrand Students

Source: (University of the Witwatersrand, 2010)

4.1.6 Sample Frame

A sample frame refers to the researched environment and the subjects used in a study (Yang et al., 2006). The sample used for purposes of the research was obtained from registered students of the University of the Witwatersrand. A random sample of 380 students who study at the University of the Witwatersrand, comprising those aged 18 and upwards, was used to collect data for this research, because these consumers would have the ability to make purchasing decisions independently, and are well suited to answer the questionnaire as it asks about purchasing behaviour.

4.1.7 Sample Size

Sample size determination is crucial in planning a statistical study, and is challenging as well (Length, 2001). The sample size was determined using the Raosoft_® sample size calculator. Raosoft takes into account the following factors; the margin of error, the confidence level, the population size and the response distribution. Default settings on Raosoft_® were utilised to calculate sample size and had a margin of error of 5%, confidence level of 95%, population size of 30 833, and a response distribution of 50 percent. Raosoft then calculated that the sample size needed for the survey would be 380 respondents.

4.1.8 Sample Method

The sampling method most appropriate for this research was probability sampling, as it provides every unit within the population an equal chance to be sampled (Gaplin, 2011; Daniel, 2011). The probability sampling method employed was the simple random sampling method. Daniel (2011) proposed six steps in selecting a simple random sample, which include, first: defining the target population; second: identifying a sampling frame of the target population; third: evaluating the sampling frame for under-coverage, coverage, multiple coverage and clustering and to make adjustments accordingly; fourth: assigning a unique number to each element in the frame; fifth: determining sample size, and lastly, the sixth step, which is to randomly select the targeted number of population elements.

4.2 QUESTIONNAIRE DESIGN (MEASUREMENT INSTRUMENT)

In any study, the theoretical constructs that are measured are the determining factors for the choice of measurement methodology (Fagarasanu and Kumar, 2002). The primary data collection instrument used for this dissertation was a research questionnaire. The questionnaire was self-administered to willing respondents at the University of the Witwatersrand, who were registered students of the University. The questionnaire comprised of five sections A, B, C, D and E. Section A asked the respondent about his or her biographical information, such as gender and age. Sections B, C, D and E employed a 7 point likert scale to ask the respondents questions that were based on the research model. Section B asked the respondents questions regarding their perceived ease or difficulty (perceived behavioural control) in the purchasing of counterfeit products. Section C asked after respondents' attitudes towards the economic benefits of purchasing counterfeit products. Section D asked about the respondents' perceptions of the relationship between price and quality of products (price-quality inference of counterfeit products). Section E asked the respondents questions about their intention to purchase counterfeit products.

questionnaire was designed based on the study's research model. Table 6 below illustrates the

questions that were adapted from published academic articles taken from accredited journals,

that were used to develop the research questionnaire for the current study.

Table 6: Adapted and Original Measurement Items

Research Construct		Adapted Measurement Item	Original Measurement Item
Perceived behavioural control	PBC 1	For me to purchase a counterfeit product over the next two weeks would be difficult.	'For me to perform behaviour over the next 2 weeks would be difficult' (disagree completely/agree completely)
	PBC 2	If I wanted to, I would not have problems in succeeding to purchase a counterfeit product over the next two weeks.	'If I wanted to, I would not have problems in succeeding to <i>perform behaviour</i> over the next 2 weeks' (disagree completely/agree completely)
	PBC 3	I have full control over purchasing counterfeit products over the next two weeks.	'I have full control over <i>performing behaviour</i> over the next 2 weeks' (disagree completely/agree completely).
	PBC 4	It is completely up to me whether or not I purchase a counterfeit product over the next two weeks.	'It is completely up to me whether or not I <i>perform</i> <i>behaviour</i> over the next 2 weeks' (disagree completely/agree completely).
	PBC 5	It is first and foremost up to me whether or not I purchase a counterfeit product over the next two weeks.	'It is first and foremost up to myself whether or not I <i>perform</i> <i>behaviour</i> over the next 2 weeks' (disagree completely/agree completely).
	PBC 6	How easy or difficult would it be for you to purchase a counterfeit product over the next two weeks?	'How easy or difficult would it be for you to <i>perform behaviour</i> over the next 2 weeks' (very difficult/very easy).
	PBC 7	How confident are you that you could purchase a counterfeit product over the next two weeks?	'How confident are you that you could <i>perform behaviour</i> over the next 2 weeks' (completely unconfident/ completely confident);
	PBC 8	If you actually tried, how likely is it that you would succeed in purchasing a counterfeit product over the next two weeks?	'If you actually tried, how likely is it that you would succeed to <i>perform behaviour</i> over the next

			2 weeks' (very unlikely/very likely)
	PBC 9	How much control do you feel over purchasing counterfeit products over the next two weeks?	'How much control do you feel over <i>performing behaviour</i> over the next 2 weeks' (no control at all/complete control).
			Source: Kraft, Rise, Sutton and Røysamb (2005)
Attitudes towards economic benefits of purchasing counterfeit products	ATT 1	I buy counterfeit products if I think genuine designer products are too expensive.	I buy counterfeit products if I think genuine designer products are too expensive.
	ATT 2	I buy counterfeit products if I cannot afford to buy designer products.	I buy counterfeit products if I cannot afford to buy designer products.
	ATT 3	I buy counterfeit products without hesitation if I have a chance to buy the counterfeits.	I buy counterfeit products without hesitation if I have a chance to buy the counterfeits.
	ATT 4	I buy counterfeit products, instead of the designer products, if I prefer specific brands.	I buy counterfeit products, instead of the designer products, if I prefer specific brands.
	ATT 5	I boast about counterfeit products as if they are the genuine brand products.	I boast about counterfeit products as if they are the genuine brand products.
	ATT 6	I usually purchase counterfeits when it is difficult to distinguish between the counterfeits and the genuine products.	I usually purchase counterfeits when it is difficult to distinguish between the counterfeits and the genuine products.
			Source: Lee and Yoo (2009)
Price-quality inference of counterfeit products	PQ1	Generally speaking, the higher the price of a product, the higher the quality.	Generally speaking, the higher the price of a product, the higher the quality.
	PQ2	The price of a product is a good indicator of its quality.	The price of a product is a good indicator of its quality.
	PQ3	You always have to pay a bit more for the best.	You always have to pay a bit more for the best.
			Source: De Matos, Ituassu and Rossi (2007)
Purchase intention of counterfeit	PI 1	The likelihood of me purchasing a counterfeit product is	The likelihood of purchasing this product is

products			
	PI 2	My willingness to buy a counterfeit product is	My willingness to buy the product is
	PI 3	The probability that I would consider buying a counterfeit product is	The probability that I would consider buying the product is
	PI 4	At the price shown, the chances of me purchasing a counterfeit product are	At the price shown, I would consider buying the product
	PI 5	If I were going to buy a product, I would consider buying a model at the price shown	If I were going to buy this product, I would consider buying this model at the price shown
			Source: Dodds, Monroe and Grewal (1991)

4.3 DATA COLLECTION

4.3.1 Data Collection Techniques

The data collection technique that was used for purposes of this research included primary and secondary data collection techniques. Primary data refers to data that is collected specifically for the purpose of the investigation at hand (Churchill and Iacobucci, 2002). Secondary data, on the other hand, are statistics not gathered for the immediate study at hand but for some other purpose (Churchill and Iacobucci, 2002). The survey questionnaire was distributed to the students of The University of the Witwatersrand. To collect data from participants, a comprehensive questionnaire was developed, whereby random students from the University of the Witwatersrand were selected to answer questions on counterfeit products purchasing.

Primary Research

Primary research is usually conducted when a problem does not have a great deal of published information (Driscoll, 2011). Primary research was conducted by self-administering research questionnaires to a sample of 380 respondents, who were, as mentioned, registered students of the University of the Witwatersrand.

Secondary Research

Baines, Fill and Page (2011) defined secondary research as also referred to as desk research as a technique used to collect data previously collected for other intentions or purposes than that of the research at hand. Accredited academic journals that spoke to the current research were consulted for purposes of this study. Relevant theory was also obtained through secondary researchm and the sources used included Google ScholarTM, the data bases available on the University of the Witwatersrand library website, as well as textbooks available in the libraries of the University of the Witwatersrand.

4.4 DATA ANALYSIS

4.4.1 Structural Equation Modeling (SEM)

Structural equation modeling (SEM) was engaged in the study for the purpose of analysing data. SEM has recently become a revered statistical technique to test theory in several fields of knowledge (Hair, Anderson, Tatham, and Black, 1998; Schumacker and Lomax, 2004; Nusair and Hua, 2010). Structural Equation Modeling (SEM) was applied so as to examine the hypothesised relationship in the research model (Liao and Hsieh, 2013). Qureshi and Kang (2015) defined structural equation modeling as a multivariate statistical technique, primarily engaged when studying relationships between latent variables (or constructs) and observed variables that constitute a model. SEM is a technique of multivariate statistical analysis, with the ability to measure the underlying latent constructs identified by factor analysis, and evaluating the paths of the hypothesised relationships between the constructs

(Nusair and Hua, 2010). According to He, Gai, Wu and Wan (2012) and Hair (2014), SEM is considered to be similar to regression analysis, but is more predominant, as it evaluates the causal relationships among constructs, while at the same time taking measurement error into consideration. The advantages of SEM can be summarised as follows; firstly it allows for the approximation of a series and multiple regression equations simultaneously (Nusair and Hua, 2010); secondly, it has the capability to incorporate latent variables into the analysis, and accounts for measurement errors in the approximation process (Hair et al., 1998, Nusair and Hua, 2010), and finally, SEM is a statistical approach that establishes measurement models and structural models to address intricate behavioural relationships (Nusair and Hua, 2010; Washington, Karlaftis and Mannering, 2003).

The SEM analysis was carried out employing a two-phase approach (Anderson and Gerbing, 1998; Hair et al., 1998). In the first phase, a confirmatory factor analysis was employed to measure the sufficiency of the measurement model (Nusair and Hua, 2010). Both construct reliability and item reliability were tested (Nusair and Hua, 2010). After establishing that the scale was reliable, the construct validity using convergent and discriminant validity was checked before the measurement model was evaluated and completed (Nusair and Hua, 2010). In the second phase of SEM, the structural model was assessed (Nusair and Hua, 2010). The overall model fit in both measurement and structural models was assessed utilising goodness-of-fit indices including c/df ratio, CFI, NFI, NFI, RFI, IFI and RMSEA (Hair et al., 1998; Jöreskog and Sörbom, 1993; Schumacker and Lomax, 2004; Nusair and Hua, 2010).

4.5 DATA ANALYSIS APPROACH

Data analysis is a statistical process in which raw data is prepared and structured so that valuable information can be extracted from it (Ullah, 2010). Firstly, the collected data was coded in an Excel spread sheet, before analysis. To gain comprehension of the attributes of

each variable, descriptive statistics analysis were utilised and indicated by the mean and standard deviation of each factor. The researcher had the full responsibility of analysing the data. The Statistical Package for the Social Sciences (SPSS) and Analysis of Moment Structures (AMOS) were employed to analyse the data. SPSS was used for Descriptive statistics, and AMOS was used to conduct confirmatory factor analysis and path modeling in order to check for reliability and validity of the data, as well as model fit of the data.

4.5.1 Data Processing and Analysis

Trochim (2000) pointed out that most social research involves the data analysis that occurs in three main stages, which are usually carried out in the following order:

- 1. Cleaning and sorting out the data for analysis (data preparation)
- 2. Describing the data (descriptive statistics)
- 3. Testing hypotheses and models (inferential statistics)
- 4. Structural equation modeling
 - 4.1 Confirmatory factor analysis (CFA)
 - 4.2 Path modeling (PM)

4.5.1.1 Data Cleaning and Coding

Before the data collected from study can be analysed, certain checks for the legitimacy of the data have to be conducted, and the researcher has to take this into consideration, so that if incorrect data was entered on the excel spread sheet it can be removed. Trochim (2000) has stated that immediately after receiving the collected data the researcher must screen it for accuracy. This will enable the researcher to identify any errors the sample might have. Trochim (2000) has added that the following questions had to be asked by the researcher in order to successfully check for discrepancies and inconsistencies:

• Are the responses written clearly?

- Did the respondent answer all important questions?
- Did the respondent complete the questionnaire?
- Does the questionnaire contain all the relevant contextual information, such as data, time, place and the researcher's details?

4.5.1.2 Descriptive Statistics

Trochim (2000) has defined descriptive statistics as information that is used to describe the basic characteristics of the data in the study. Descriptive statistics are used to provide behavioural patterns of respondents in general (Hsu and Shine, 2007). Descriptive statistics involve simple summaries about the samples and the dimensions of the data. The descriptive statistics could take the form of pie charts, or tables, that show the basic data of the main components of the study, for example demographic or biographical data. In the current study descriptive statistics have explored the demographic characteristics of the research data. The total number of participants were mentioned, and the distribution of gender, age, marital status, educational level and purchasing behaviour of participants was explored.

4.5.2 Reliability and Validity of Measurement Scales

Reliability and validity both relate to the logic and accuracy of a test (Wilckens, 2010). Reliability is the degree to which the instruments consistently measure what the instrument is supposed to measure (Dusick, 2011), and this is done numerically, with a coefficient greater than a Cronbach's alpha of 0.6, suggesting that the reliability of the research instrument is marginally acceptable.

A validity test is undertaken to identify whether an instrument of measurement tool has performed its intended measurement function (Budiman and Wijaya, 2014). Budiman and Wijaya (2014) have pointed out that to know the data consistency and accuracy collected from instrument use, a validity test is used by using the correlation of product moment (Pearson). Reliability requires better comparable experiments, while validity asks the question as to whether the experiment is tailored to appropriately answer the questions being asked, i.e. if the experiment is valid in logical terms (Wilckens, 2010). In particular, the factor loadings, the Cronbach's alpha values and Composite Reliability (CR) values were determined using the Statistical Package for the Social Sciences (SPSS) 22.0 and Analysis of Moment Structures (AMOS) 22.0 software, in order to assess measurement items' reliability. Convergent and discriminant validity of the research constructs were determined by checking the inter-correlation between the research constructs, comparing the average variance extracted (AVE), and shared variance (discriminate validity). As for convergent validity, the item total correlation values, item loading and average variance extracted were utilised as indicators.

4.5.3 Tests of Measures and Accuracy Analysis Statistics

After demographic data analysis, the reliability and validity of the measuring scales need to be assessed so as to ensure valid data analyses. This was principally important for this study, as a few of the scales had been modified in order to adapt to the particular research context. The total error of a measurement entails of systematic errors, which involve with validity test, and random errors, which are measured by reliability test (Aaker, 2007).

4.6 RELIABILITY TESTS

Three methods, namely Cronbach's alpha test (Cronbach α), the composite reliability test (CR) and average variance extracted (AVE) test, were used in this thesis to check on the reliability of the research measures. Reliability refers to the extent to which the instruments consistently measure that which they ought to measure (Dusick, 2011) and this is done numerically, with a coefficient greater than a Cronbach's alpha of 0.6, suggesting that the reliability of the research instrument is marginally acceptable. Boyle (1991) stated that the term 'internal consistency' has been used expansively in classical psychometrics to refer to

the reliability of a scale based on the degree of within-scale item inter-correlation, as measured by the split-half method, or more adequately, by Cronbach's alpha. According to Dunn, Baguley and Brunsden (2013) the coefficient alpha is the most commonly used measure of reliability, and certainly of internal consistency reliability reported in psychological research.

4.6.1 Cronbach's Alpha Test

The Cronbach's alpha was developed by Lee Cronbach in 1951 to offer a measure of the internal consistency of a test or scale, and is expressed as a number between 0 and 1 (Tavakol and Dennick, 2011). In this study, the internal reliability of each construct was assessed using the standardised Cronbach's coefficient alpha. Cronbach's coefficient α is one of the most common internal consistency approaches (Dunn, Baguley and Brunsden, 2013). According to Chinomona (2011), a higher level of Cronbach's coefficient alpha indicates a higher reliability of the measurement scale.

Table 7 below is an illustration of reliability test criteria, description for each criteria and the acceptance level for each criteria.

Reliability test	Description	Acceptable level	Source
criteria			
Cronbach's Alpha	Coefficient for determining	Value must be equal to or	Nunnally and
	internal consistency of items	greater than 0.7	Bernstein (1994)
Composite	Method used to measure internal	Value must be equal to or	Hair <i>el al</i> (2006)
reliability (CR)	consistency of the measurement	greater than 0.7	
	model		

 Table 7: Reliability Test Criteria, Description and Acceptable Level

Average Variance	Method used to measure	Value must be equal to or	Fraering and Minor
Extracted (AVE)	reliability and validity of items	greater than 0.5	(2006)

For purposes of this research, the Cronbach's Alpha, Composite reliability (CR) and Average Variance Extracted (AVE) were conducted so as to assess the reliability of the measures.

4.6.2 Composite Reliability

According to Yang and Lai (2010) in reliability analysis, an acceptable CR value must exceed 0.7. The internal reliability of each construct was also evaluated using the Composite Reliability (CR) index test. It is calculated using the following formula:

(CR): $CR_{\eta} = (\Sigma \lambda yi)^2 / [(\Sigma \lambda yi)^2 + (\Sigma \epsilon i)]$

Composite Reliability = (square of the summation of the factor loadings) / {(square of the summation of the factor loadings) + (summation of error variances)}.

The resultant coefficient is similar to that of Cronbach's α. The threshold for Composite Reliability (CR) index of 0.5 for basic research and 0.6 for exploratory research are suggested by Nunnally (Chinomona, 2011). The value is later adjusted to 0.7, and is recommended by Hair, Anderson, Tatham and Black (2006). Following the formula provided above to calculate Composite Reliability, the results that were found are illustrated in Table 10 above. According to the broader literature, a Composite Reliability index that is greater than 0.7 depicts an adequate internal consistency of the construct (Hair et al., 2006).

4.7 VALIDITY TESTS

Validity refers to the degree to which a test or instrument measures the attributes that it is supposed to measure (Easterby-Smith, Thorpe and Lowe, 2002). Its primary concern is with the accuracy of measurement. This section of the chapter is concerned with testing construct validity.

4.7.1 Construct Validity

Construct validity is a type of validity that has attracted the greatest amount of attention in social science (Churchill, 1979). To establish this type of validity, two categories of construct validity normally need to be determined, namely: convergent validity and discriminant validity.

4.7.2 Convergent Validity

According to Sarstedt, Ringle, Smith, Reams and Hair (2014), convergent validity measures the degree to which a construct comes together in its indicators by explaining the items' variance. Convergent validity is considered sufficient when the AVE value of each construct exceeds 0.5 (Yang and Lai, 2010). Peter (1981) stated that ideally, an item is expected to be related to other items that measure the same constructs (convergent validity), but to differ from items which measure different constructs (discriminant validity).

4.7.3 Discriminant Validity

Discriminant validity refers to the degree to which a measure is distinct from other measures, i.e. it shows heterogeneity between different constructs (Malhotra, 1996). According to Fornell and Larcker (1981), discriminant validity can be assessed using the average variance extracted (AVE). To confirm discriminant validity, the average variance extracted for each construct should be greater than the squared correlations between the construct and all other constructs in the model (Nusair and Hua, 2010).

4.7.3.1 Inter-construct Correlation Matrix

One of the methods used to check on the discriminant validity of the research constructs was the evaluation of whether the correlations among latent constructs were less than 1.0.. A correlation value between constructs of less than 0.7 is recommended in the empirical literature to confirm the existence of discriminant validity (Bagozzi and Yi, 1991; Nunnally and Bernstein, 1994).

4.7.3.2 Average Variance Extracted (AVE) and Shared Variance (SV)

Discriminant validity was also established by checking if the average variance extracted (AVE) was greater than the highest shared variance (HSV) (Fornell and Larcker, 1981).

4.8 Model Fit/Acceptability

According to Schumacker and Lomax (2004), CFA and SEM fit indices have no sole statistical test of significance that identifies a correct model given the sample data, especially alternative models can exist that yield exactly the same data to model fit. It is recommended that various model fit criteria be used in combination to assess model fit as global fit measures (Hair et al., 1992).

4.8.1 Confirmatory Factor Analysis

Confirmatory factor analysis is an analytical tool that allows the researcher to explore hypotheses about what constructs the test in question is measuring and provides an empirical basis for clinical interpretation (Burton, Ryan, Axelrod, Schellenberger and Richards, 2003). A confirmatory factor analysis was performed to obtain the standard regression weights. Model fit indicators such as Chi-square/degrees of freedom, Goodness of Fit Index (GFI), Normed Fit Index (NFI), Incremental Fit Index (IFI), Tucker-Lewis Index (TLI), Composite Fit Index (CFI) and the Random Measure of Standard Error Approximation (RMSEA) will be used to assess the model fit.

This chapter used an eight model fit criteria to test the overall fit of the model. Table 8 on the following page presents the acceptable model fit criteria.

Table 8:]	Model Fit	Criteria	and	Acceptable	Fit	Level
------------	-----------	----------	-----	------------	-----	-------

Model fit	Description	Acceptable level	Source
criteria			
Chi-square (χ^2)	Method used to assess the general fit of the	Value must be below 3	Chinomona
/ D F)	model		(2011)
			(2011)
Goodness of Fit	The (GFI) is the degree of fit between the	Ranges between 0 and 1,	Baumgartner
Index (GFI)	hypothesized model and the observed	with a cutoff value of 0.9	and Hombur
	covariance matrix.		(1996)
Normed Fit	The (NFI) evaluates the discrepancy between	Value must be greater than	Bentler and
Index (NFI)	the chi-squared value of the hypothesised	0.9	Bonett (1980)
	model and the chi-squared value of the null		
	model.		
Tucker-Lewis	The TLC utilises simpler models and is known	Value must meet or exceed	Hooper,
Index (TLI)	to address the issue of sample size associated	0.9	Coughlan and
	with NFI.		Mullen (2008);
			Chinomona
			(2011)
Incremental Fit	The IFI's purpose is to correct the issue of	Value must meet or exceed	Bollen, (1989),
Index (IFI)	parsimony and sample size related to NFI.	0.9	Chinomona
			(2011)
Comparative Fit	The (CFI) assumes that all latent variables are	Value must meet or exceed	Chinomona
Index (CFI)	uncorrelated and compares the sample	0.9	(2011),
	covariance matrix with the null model.		Hu and Bentler
			(1999), Hooper

			<i>et al</i> (2008)
Relative Fit	The (RFI) compares the chi-square for the	Value must meet or exceed	McDonald and
	The (IIII) compares the emisquate for the	value must meet of enceed	ine on and
Index (RFI)	hypothesised model to one from a "null", or	0.9	Ho (2002)
	"baseline" model.		
Root mean	The (RMSEA) informs how well the model,	Value must fall below 0.05	Byrne (1998)
square error of	with indefinite but optimally selected	and 0.08	
approximation	parameter estimates, would fit the populations		
(RMSEA)	covariance matrix (Byrne, 1998).		

4.8.1.1 Chi-square (χ^2 /DF)

According to Nevitt and Hancock (2000) the chi square fit statistic tests a hypothesis of precise fit of the proposed model in the population. A chi-square value below three (3) is considered to be an acceptable model fit as suggested by (Chinomona, 2011).

4.8.1.2 Goodness-of-fit Index (GFI)

The Goodness of Fit is one of many criterion values for indicating satisfactory model fit suggested by researchers (Cheung and Rensvold, 2002). GFI varies from 0-1, but theoretically can yield meaningless negative values. Through general consensus GFI should be equal to or greater than 0.90 to accept the model (Bollen, 1990).

4.8.1.3 Normed Fit Index (NFI)

The Normed Fit Index (NFI) assesses the inconsistency between the chi-squared value of the hypothesised model and the chi-squared value of the null model (Bentler and Bonett, 1980). It is generally agreed upon that NFI values below 0.90 indicate a need to re-specify the model (Hu and Bentler, 1999).

4.8.1.4 Tucker-Lewis Index (TLI)

The Tucker-Lewis Index (TLI) uses simple models and is known to address the issue of sample size associated with The Normed Fit Index (NFI). Recommended value must meet or exceed 0.9 (Hooper et al., 2008; Chinomona, 2011).

4.8.1.5 Incremental Fit Index (IFI)

Bollen (1989) introduced the IFI in order to address the issue of parsimony and sample size, which was known to be associated with the NFI. Chinomona (2011) stated that the recommended IFI should be equal to or greater than 0.9 in order to accept the model.

4.8.1.6 Comparative Fit Index (CFI)

According to Gatignon (2010) the comparative fit index (CFI) analyses the model fit through assessing the discrepancy between the data and the hypothesised model. The CFI is a revised version of the normed fit index (NFI), which is responsible for the sample size (Byrne, 1998). The CFI also addresses sample size issues normally associated with the chi-square test and the normed fit index (Bentler, 1990), and functions well, even when the sample size being used for the study is small (Tabachnick and Fidell, 2007). According to Hu et al. (1999) and Chinomona (2011), a value equal or greater than 0.9 is an indication of acceptable model fit.

4.8.1.7 Root Mean Square Error of Approximation (RMSEA)

The root mean square error of approximation (RMSEA) fit index was introduced by Steiger and Lind (1980) for evaluating covariance structure models (Steiger, 1998). It reduces problems and inconsistencies commonly found in testing models with large sample sizes, and have therefore become a helpful tool for guiding complex judgments about model utility, rather than functioning as a replacement for such judgements (Steiger, 1998). A good model fit if RMSEA is considered to be less than or equal to 0.5 and an adequate fit if RMSEA is less than or equal to 0.8 (Chinomona, 2011).

4.8.2 Path Modeling

Path modeling describes the relationships between observed or measured variables and theoretical constructs (Roche, Duffield and White, 2011) and tests the structural paths of the conceptualised research model. Once the model fit was assessed using confirmatory factor analysis (CFA), this study proceeded to perform path modeling using the AMOS 22 software package. The structural equation modeling (SEM) technique demonstrates and tests the theoretical underpinnings of a proposed study, and the significance of the relationships between models constructs. SEM stipulates a technique where separate relationships are allowed for each set of dependent variables, and provides an estimation technique for a series of separate multi-regression equations to be estimated concurrently. It further contains two mechanisms, namely the structural model, which is the path where independent and dependent variables are linked, and the measurement model, which enables this study to use several indicators for a single independent variable.

4.9 ETHICAL CONSIDERATIONS

An ethics clearance form was applied for and obtained, as it was one of the requirements of the research. This is the case because a human element was involved in the study, and therefore measures had to be put in place to ensure the protection of peoples' privacy as well as the reputation of the University of the Witwatersrand. The University of the Witwatersrand ethics committee made the decision to grant ethical clearance for the research and, if any fabrication, falsifying or misrepresenting of research data occurred, the committee had full authority to take appropriate action. No participant was forced to take part in the study, and no incentives were used to lure participants into taking part. Participants were free to pull out of the study at any given point. All the procedures as far as ethical conduct throughout the research were adhered to. All the collected data was kept as confidential, and none was used for any other purposes other than those of the research.

4.10 SUMMARY OF CHAPTER 4

Chapter 4 explored the research methodology and design used in this study, and defined this as a quantitative study. A self-administered questionnaire was used in order to assess how customers perceive the purchasing of counterfeit products. In total, 393 questionnaires were collected and 380 usable questionnaires were captured, cleaned and analysed. The chapter was divided into 12 sections. The first section was the introduction, the second, a discussion of quantitative research, in the third, a sampling design was presented, and the fourth explores the sample size. In the fifth section, the sample method was also discussed. Chapter 5 explores the data analysis process as well as discusses the results of the study. Chapter 5 also includes the interpretation of the results and ends with a summary of the entire chapter. Figure 7 and 8 on the following page illustrate the research methodology and data analysis approach respectively that was used for this dissertation.



Figure 7: Diagrammatic Representation of Research Methodology

Figure 8: Diagrammatic Representation of Data Analysis Approach





Figure 9: Diagrammatic Representation of Chapter 5

CHAPTER 5: DATA ANALYSIS AND DISCUSSION OF RESULTS

5.1 INTRODUCTION

Chapter 4 discussed the methods and techniques that were used to obtain the findings that will be presented in this chapter. This chapter presents and discusses the findings that were obtained through empirical investigation. This chapter presents statistical analysis of data that was collected through the data collection tool (research questionnaire). To analyse the data, the Statistical Package for the Social Sciences (SPSS) was utilised. In this chapter, descriptive statistics were discussed, and the reliability of all the constructs in the model used to develop the questionnaire was also discussed. Structural Equation Modeling was also conducted, where Confirmatory Factor Analysis and Path Modeling were conducted. Confirmatory Factor Analysis (CFA) was conducted so as to check for Model Fit, Reliability and Validity of the scales used in the research questionnaire. To check the validity of the scales, shared variance was compared to average variance extracted (AVE). Path Modeling (PM) was conducted to check for model fit, and to test the hypothesis of the study.

Chapter 5 will commence by exploring the data screening process, followed by the presentation of data analysis procedures employed by the current study. Thereafter, the sample description is provided. Following the sample description, a test of measures and accuracy analysis statistics is provided. This section of the chapter mainly tests for the measures' reliability and validity, using a variety of methods to ascertain accuracy. For measure reliability, the Cronbach's Alpha, the Composite Reliability (CR) value, and the Average Value Extracted (AVE) were used; whilst for validity, factor analysis was used to check convergent validity and correlation matrix and chi-square, where CFA difference was used to check for evidence of discriminant validity. This section is followed by a presentation of the research models fit. Numerous indicators such as the chi-square value, Goodness of Fit Index (GFI), Normed Fit Index (NFI), Incremental Fit Index (IFI), Comparative Fit Index

(CFI) and Root Mean Square Error of Approximation (RMSEA), were used to ascertain whether the research model fit the data. In conclusion, a summary of Chapter 5 is provided.

5.2. DATA SCREENING

After data collection, the researcher implemented the data screening process proposed by Malhotra (1993) and Churchill (1999), which was done to ensure data was cleaned before conducting any further statistical analysis. Screening the data is the initial step towards obtaining insight into the characteristics of the data. It is crucial to ensure the accuracy of data entries and assessment of outliers, before proceeding to analyse summary statistics for the survey responses. The major analytical tasks in the data screening process include questionnaire checking, editing, coding, and tabulation. Using SPSS, each data field was tested for mean and standard deviation, so as to detect any typographical errors and possible outliers. Data was cleaned after errors in data entry were rectified.

5.3. DATA ANALYTICAL PROCEDURES

To analyse the empirical data, several statistical methods were employed. Firstly, coefficient alpha and adjusted item-to-total correlations were used in assessing the internal consistency of each construct. Data was analysed using SPSS. For the assessment of final measures, confirmatory factor analysis was performed using the AMOS 22. Statistical procedures used to validate measures involved the assessment of items and scale reliability, convergent and discriminant validity. Details of structural equation modeling will be analysed in this chapter, along with the interpretation of results. Figure 10 demonstrates the procedures of statistical analysis, as well as the key tasks that will be undertaken in Chapter 5.



Figure 10: Statistical Analysis Procedures

5.4 DESCRIPTIVE STATISTICS

Kneale and Santy (1999) have stated that any study should commence by explaining the demographic or descriptive traits of the sampled population, and that it ought to present this in a comprehensible way. The purpose of descriptive statistics is to search for patterns, to put together and present a set of data describing the characteristics of the sample so as to make comparisons (Hsu and Shine, 2007). Descriptive statistics involve simple summaries about the samples and the dimensions of the data. The descriptive statistics could take the form of pie charts or tables, showing the basic data of the main components of the study for example demographic or biographical data.

5.4.1 Sample Description

Table 7 presents the profile of the participants. The University of Witwatersrand, which has a total population of 30 833 (University of the Witwatersrand, 2010), was used as the sample frame, and only registered students of the University of the Witwatersrand were surveyed.

The profile indicates that the proportion of males to females was almost evenly split, with males representing 49,2% of the total sample and females accounting for 50,8% of the total sample. Most of the respondents were from the age group of (18-19 years) and the age group with the least number of respondents (26 years and older) was the represented by 26,6% of the total sample. Most of the respondents were single, a group comprising 94, 2% of the total sample, as compared to the married respondents, who only comprised 5,8% of the total sample. The results showed that most of the respondents indicated a high school education as their highest qualification, indicated by 47,1% of the total sample. All respondents were students, where 62.1% indicated that their purchase frequency was mostly on a monthly basis. Most of the respondents indicated that they spent less than R250 on luxury goods, which was represented by 104 out of the total 380 respondents. Table 9 below presents a profile of the participants.

Gender	Frequency	Percentage
Male	187	49,2 %
Female	193	50,8 %
Total	380	100 %
Age	Frequency	Percentage
18-19	158	41,6 %
20-25	121	31,8 %
26+	101	26,6 %
Total	380	100 %
Marital Status	Frequency	Percentage
Married	22	5,8 %

Table 9: Sample Demographic Profile

Single	358	94,2 %
Total	380	100%
Level of Education	Frequency	Percentage
High School	179	47,1 %
Diploma	27	7,1 %
Degree	95	25,0 %
Post Graduate	79	20,8 %
Total	380	100%
Occupation	Frequency	Percentage
Student	380	100,0 %
Total	380	100%
Purchase Frequency	Frequency	Percentage
Purchase Frequency Monthly	Frequency 236	Percentage 62,1 %
Purchase Frequency Monthly Seasonally	Frequency 236 65	Percentage 62,1 % 17,1 %
Purchase Frequency Monthly Seasonally Annually	Frequency 236 65 79	Percentage 62,1 % 17,1 % 20,8 %
Purchase Frequency Monthly Seasonally Annually Total	Frequency 236 65 79 380	Percentage 62,1 % 17,1 % 20,8 % 100%
Purchase FrequencyMonthlySeasonallyAnnuallyTotalAmount of Money Spent	Frequency 236 65 79 380 Frequency	Percentage 62,1 % 17,1 % 20,8 % 100% Percentage
Purchase FrequencyMonthlySeasonallyAnnuallyTotalAmount of Money SpentLess than R250	Frequency 236 65 79 380 Frequency 104	Percentage 62,1 % 17,1 % 20,8 % 100% Percentage 27,4 %
Purchase FrequencyMonthlySeasonallySeasonallyAnnuallyTotalAmount of Money SpentLess than R250Between R250- R500	Frequency 236 65 79 380 Frequency 104 82	Percentage 62,1 % 17,1 % 20,8 % 100% Percentage 27,4 % 21,6 %
Purchase FrequencyMonthlySeasonallySeasonallyAnnuallyTotalAmount of Money SpentLess than R250Between R250- R500Between R501- R1000	Frequency 236 65 79 380 Frequency 104 82 96	Percentage 62,1 % 17,1 % 20,8 % 100% Percentage 27,4 % 21,6 % 25,3 %
Purchase FrequencyMonthlySeasonallySeasonallyAnnuallyTotalLess than R250Between R250- R500Between R501- R1000More than R1000	Frequency 236 65 79 380 Frequency 104 82 96 98	Percentage 62,1 % 17,1 % 20,8 % 100% Percentage 27,4 % 21,6 % 25,3 %



Figure 11: Pie Chart Illustrating the Distribution of Males and Females in the Sample

The Pie Chart in Figure 11 above illustrates the gender distribution of male to female of the sample. It can be observed that males accounted for 50,79% of the total sample number, and that females accounted for 49,21% of the total sample. Below Figure 12 presents a bar graph illustrating the educational levels of the respondents.

Figure 12: Bar Graph illustrating Educational Levels of the Respondents



As can be seen in Figure 12 on the previous page, most of the respondents stated high school education as their highest education level, indicated by the range between 150 and 200 respondents, with the least number of respondents indicating a diploma as their highest education level, as indicated by 7.1% of the total sample. Figure13 shows a bar graph illustrating the amount of money spent by respondents on luxury products.



Figure 13: Bar Graph Illustrating Amount of Money Spent by Respondents

As can be seen in Figure 13 above, most of the respondents spend less than R250, which is indicated by a value above 100 respondents, where those respondents that spent between R250 and R500 had the lowest representation, only accounting for value close to 80 out of the total 380 respondents. Figure 14 on the following page illustrates the purchase frequency of respondents.


Figure 14: Bar Graph illustrating Purchase Frequency of Respondents

As can be seen in Figure 14 above, most of the respondents purchase luxury products on a monthly basis, which was represented by a range of 200 to 250 of the total 380 respondents. Thereafter, respondents who purchase luxury products annually, had the second highest representation, and were represented by a range between 50 to 100 respondents. Finally, the least number of respondents indicated that they purchase luxury products seasonally; and these were also around 50 to 100 respondents, but noticeably lower than those who purchase luxury products annually. The following section discusses the results of the questionnaire.

5.4.2 Questionnaire Results

All research constructs were measured on a seven-point Likert-type scale. There were a total number of 380 respondents. The first variable "Perceived Behavioural Control (PBC)" was measured with nine items, ranging from PBC1 to PBC9. The second variable, namely "Attitudes towards economic benefits of purchasing counterfeit products (ATT)" was measured with six items, ranging from ATT1 to ATT6. The third variable "price-quality

inference of counterfeit products (PQ)", was measured with three items ranging from PQ1 to PQ3. The forth variable, "purchase intention of counterfeit products (PI)", was measured with five items ranging from PI1 to PI5. Having measured the construct "perceived behavioural control (PBC)" with nine measurement items, results indicate that most of the respondents were neutral in their responses. This is evident in the following neutral responses for PBC: PBC1 obtained 173 (45.5%); PBC2 obtained 210 (53.5%); PBC3 obtained 195 (51.3%); PBC4 had 208 (54.7%); PBC5 achieved 177 (46.6%); PBC6 reached 196 (51.6%); PBC7 reached 199 (52.4%); PBC8 obtained 200 (52.6%); PBC9 obtained 202 (53.2%). The variable "attitudes towards economic benefits of purchasing counterfeit products (ATT)" showed that most of the respondents were leaning towards neutrality, or were agreeing with the statements/questions. For example, ATT2 had 217 (57.1%) neutral responses, and ATT1 had 115 (30.3%) respondents. Out of the possible 380 agreeing with the statement, this was significantly more than the other responses for the same statement, such as one respondent (0.3%) and four respondents (1.1%) strongly disagreeing and moderately disagreeing, respectively. The other variable, "price-quality inference of counterfeit products (PQ)", proved that most of the respondents were neutral. This was indicated by 206 (54.2%) neutral responses for PQ1. However, following the neutral responses, more respondents agreed with the statements/questions, as compared to disagreeing with them. For example, PQ3 had 97 respondents (25.5%), stating that they agreed with the statement, as compared to one respondent (0.3%), who strongly disagreed with the same statement. The final variable, "purchase intention of counterfeit products (PI)" had higher responses options 4 and 5, representing "neutral" and "high" respectively as compared to options 1 and 2, representing "very low" and "moderately low". For example 196 respondents (51.6%) indicated that they were neutral on PI5, while 109 respondents (27.8%) selected "high"; as compared to the respondents who chose "very low" and "moderately low" both of which were indicated by

one respondent (0.3%), respectively. Overally, it can be noted that there were no extremes, as few respondents indicated that they strongly disagreed with any of the statements/questions, with none stating that they strongly agreed with any of the statements/questions. Table 10 below illustrates the results for the tests to be explored in greater detail in the sections to follow.

5.5 RELIABILITY TESTS

Three approaches were employed to check for reliability of research measures used for this study, namely the Cronbach's alpha test (Cronbach α), Composite Reliability test (CR) and Average Value Extracted (AVE) test. Table 10 on the following page presents the results for all three tests used to check the research measure reliability.

5.5.1 Cronbach's Alpha Test

Internal reliability of each construct was measured using the standardized Cronbach's coefficient alpha. Higher values of Cronbach's coefficient alpha represented higher reliability of the scale. Furthermore, higher inter-item correlations revealed statistical agreement among the measured items. The results of scale reliability tests are shown in Table 10 above. As can be seen, item-to-total values ranged from 0.648 to 0.948 and therefore, were above the cut-off point of 0.3 (often ≤ 0.3) endorsed by Dunn, Seaker and Waller (1994). As can also be observed on Table 10 Cronbach's alpha coefficients ranged from to 0.838 to 0.960 surpassing the 0.7 threshold recommended by (Nunnally and Bernstein, 1994).

5.5.2 Composite Reliability (CR)

Internal reliability of each construct was also evaluated using the Composite Reliability (CR) index test. According to the literature a Composite Reliability index greater than 0.7 depicts a passable internal consistency of the construct (Hair et al., 2006). The results in Table 10

indicate that composite reliability (C.R.) indexes were between 0.835 and 0.959. These values exceeded the estimate criteria used by prior literature (Hair et al., 2006).

Research Construct		Descriptive Statistics				Cronbach's Test					
		Mean Value		Standard Deviation		Item-total	α value	C.R. Value	A VE Value	Highest Shared Variance	Factor Loading
	PBC1	3,858		0,776	0,741	,709		0,959		0.520	0,653
	PBC2	3,953		0,718		,719					0,655
	PBC3	3,966	3,901	0,756		,723	0,960				0,674
	PBC4	3,926		0,730		,772			0,730		0,738
PBC	PBC5	3,863		0,794		,838					0,871
	PBC6	3,887		0,730		,941					0,991
	PBC7	3,884		0,724		,948					0,998
	PBC8	3,887		0,723		,941					0,994
	PBC9	3,887		0,719		,943					0,994
	ATT1	4,068	3,987	0,755	0,740	,733	0,889	0,886	0,833	0.601	0,760
	ATT2	4,024		0,683		,731					0,749
ATT	ATT3	3,989		0,737		,663					0,718
AII	ATT4	4,021		0,722		,730					0,788
	ATT5	3,958		0,747		,702					0,751
	ATT6	3,863		0,797		,686					0,745
	PQ1	3,976		0,721		,706		0,835	0,757	0.601	0,826
PQ	PQ2	4,029	4,017	0,713	0,712	,740	0,838				0,801
	PQ3	4,045		0,701		,657					0,748
	Pl1	3,887		0,762		,713		0,859	0,798	0.566	0,771
PI	Pl2	3,861	3,961	0,782	0,752	,707					0,746
	Pl3	3,987		0,739		,710	0,869				0,735
	PI4	3,992		0,764		,648					0,686
	PI5	4,079		0,715		,691					0,769

Table 10: Accuracy Analysis Statistics

* Scores: 1 – Disagree completely; 4 – Neutral; 7 – Agree completely.

- * Scores: 1 Very difficult; 4 Neutral: 7 –Very easy.
- * Scores: 1 Completely unconfident; 4 Neutral<u>;</u> 7 Completely confident.
- * Scores: 1 Very unlikely; 4 Neutral: 7 Very likely.
- * Scores: 1 No control at all; 4 Neutral: 7 Complete control.
- * Scores: 1 Strongly disagree; 4 Neutral; 7 Strongly agree.
- * Scores: 1 Very low; 4 Neutral: 7 Very high.
- C.R.: Composite Reliability; AVE: Average Variance Extracted; S.V.: Shared Variance.
- ^a significance level p < 0.05; ^b significance level p < 0.01; ^c significance level p < 0.001

Measurement model fits: $\chi^2/df = 2,456$; GFI= 0,901; CFI= 0,971; IFI=0,971; TLI= 0,964; RFI=0,941; NFI=

0,952; RMSEA= 0,062

5.5.3 Average Value Extracted (AVE)

The average variance extracted proved that the overall amount of variance in the indicators were accounted for by the latent construct. Values for the variance extracted greater than 0.40 demonstrated that the indicators adequately represented the latent construct. Overall, all average variance explained (AVE) values ranged from 0.730 to 0.833, thus within the marginal to acceptable threshold recommended by (Fraering & Minor, 2006). As can be seen in table 10 on the previous page, the average variance extracted (AVE) of perceived behavioural control (PBC) is 0.73 is greater that the square of the shared variance of perceived behavioural control (PBC) and attitudes towards economic benefits of purchasing counterfeit products (ATT), which is $[(0.6652)^2 = 0.425]$. This proves the existence of discriminate validity.

Table 11 below presents the calculations for Composite Reliability.

				Composite reliability (CR)						
					summation of error terms					
				(∑λYi)²	έi	∑έi	CR			
PBC1	<	PBC	0,653		0,573591					
PBC2	<	PBC	0,655		0,570975					
PBC3	<	PBC	0,674		0,545724					
PBC4	<	PBC	0,738		0,455356					
PBC5	<	PBC	0,871	57,27462	0,241359	2,432848	0,959254			
PBC6	<	PBC	0,991		0,017919					
PBC7	<	PBC	0,998		0,003996					
PBC8	<	PBC	0,994		0,011964					
PBC9	<	PBC	0,994		0,011964					
ATT1	<	ATT	0,760		0,4224					
ATT2	<	ATT	0,749		0,438999					
ATT3	<	ATT	0,718	20 24012	0,484476	2 605005	0 006170			
ATT4	<	ATT	0,788	20,54912	0,379056	2,005905	0,000470			
ATT5	<	ATT	0,751		0,435999					
ATT6	<	ATT	0,745		0,444975					
PQ1	<	PQ	0,826		0,317724					
PQ2	<	PQ	0,801	5,64063	0,358399	1,116619	0,834752			
PQ3	<	PQ	0,748		0,440496					
PI1	<	PI	0,771		0,405559					
PI2	<	PI	0,746		0,443484					
PI3	<	PI	0,735	13,74185	0,459775	2,246861	0,859472			
PI4	<	PI	0,686		0,529404					
PI5	<	PI	0,769		0,408639					

Table 11: Diagrammatic Representation of Composite Reliability Calculations

According to Yang & Lai (2010) in reliability analysis, an acceptable CR value must exceed 0.7. The internal reliability of each construct was also evaluated using the Composite Reliability (CR) index test. It is calculated using the following formula:

(CR): $CR_{\eta} = (\Sigma \lambda yi)^2 / [(\Sigma \lambda yi)^2 + (\Sigma \epsilon i)]$

Composite Reliability = (square of the summation of the factor loadings)/ {(square of the summation of the factor loadings) + (summation of error variances)}.

The results in Table 11 above indicate that composite reliability (C.R.) indexes were between 0.835 and 0.959. These values surpassed the estimate criteria used by past literature.

Using the results of the construct "Perceived behavioural control" to demonstrate, the calculation for Composite Reliability was conducted as follows:

Step1:
$$(\Sigma\lambda yi)^2 = (0.653 + 0.655 + 0.674 + 0.738 + 0.871 + 0.991 + 0.998 + 0.994 + 0.994)^2$$

= 57, 27462

Step2: $\Sigma \epsilon i = (1-0.653)^2 + (1-0.655)^2 + (1-0.674)^2 + (1-0.738)^2 + (1-0.871)^2 + (1-0.991)^2 + (1-0.998)^2 + (1-0.994)^2 + (1-0.994)^2$

= 2,432848

Step3: CR_{η} = 57, 27462 / (57, 27462 + 2, 432848)

= 0,959254

Table 12 below presents the calculations for Average Variance Extracted.

			Estimate	λyi²	Σλyi²	έi	Σἐί	Σλyi² / (Σλyi² + Σἑi)	
PBC1	<	PBC	0,653	0,426409		0,573591			
PBC2	<	PBC	0,655	0,429025		0,570975			
PBC3	<	PBC	0,674	0,454276		0,545724			
PBC4	<	PBC	0,738	0,544644		0,455356	2,432848	0,729683556	
PBC5	<	PBC	0,871	0,758641	6,567152	0,241359			
PBC6	<	PBC	0,991	0,982081		0,017919			
PBC7	<	PBC	0,998	0,996004		0,003996			
PBC8	<	PBC	0,994	0,988036		0,011964			
PBC9	<	PBC	0,994	0,988036		0,011964			
ATT1	<	ATT	0,76	0,5776		0,666378	4,07417	0,83318506	
ATT2	<	ATT	0,749	0,561001	0,68527	0,685278			
ATT3	<	ATT	0,718	0,515524		0,734235			
ATT4	<	ATT	0,788	0,620944	20,349121	0,614429			
ATT5	<	ATT	0,751	0,56400		0,681903			
ATT6	<	ATT	0,745	0,555025		0,691947			
PQ1	<	PQ	0,826	0,682276		0,534499			
PQ2	<	PQ	0,801	0,641601	5,640625	0,588348	1,809803	0,757087389	
PQ3	<	PQ	0,748	0,559504		0,686955			
PI1	<	PI	0,771	0,594441		0,64664			
PI2	<	PI	0,746	0,556516		0,69029			
PI3	<	PI	0,735	0,540225	13,741849	0,708157	3,473918	0,798212982	
PI4	<	PI	0,686	0,470596		0,778539			
PI5	<	PI	0,769	0,591361		0,650292			

 Table 12: Diagrammatic Representation of Average Variance Extracted Calculations

The average variance extracted estimate reveals the overall amount of variance in the indicators accounted for by the latent construct (Fraering and Minor, 2006). Higher values for the variance extracted estimate (greater than 0.50) reveal that the indicators well represent the latent construct (Fraering and Minor, 2006).

The formula below is used to calculate Average Variance Extracted (AVE):

$$V_{\eta} = \Sigma \lambda y i^2 / (\Sigma \lambda y i^2 + \Sigma \varepsilon i)$$

AVE = {(summation of the squared of factor loadings)/{(summation of the squared of factor loadings) + (summation of error variances)}

Using the results of the construct PBC to demonstrate, the calculation for AVE was conducted as follows:

Step 1:
$$\Sigma\lambda yi^2 = (0.653^2 + 0.655^2 + 0.674^2 + 0.738^2 + 0.871^2 + 0.991^2 + 0.998^2 + 0.994^2 + 0.994^2)$$

= 6,567
Step 2: $\Sigma\epsilon i = (1-0.653)^2 + (1-0.655)^2 + (1-0.674)^2 + (1-0.738)^2 + (1-0.871)^2 + (1-0.991)^2 + (1-0.991)^2$

 $(0.998)^2 + (1-0.994)^2 + (1-0.994)^2$

= 2,433

Step 3: V_{η} = 6,567 / (6,567+2,433)

= 0,730

The above demonstrated steps were carried out when calculating the AVE of each of the research constructs. As indicated in Table 12, a good representation of the latent construct by the item is identified when the variance extracted estimate is above 0.5 (Fraering and Minor, 2006). The results of AVE range from 0.730 to 0.833 in Table 12 thereby confirm an acceptable representation of the latent construct by the items.

Altogether, the construct reliabilities and the average variance extracted estimates suggest the scales are internally consistent. In this study, the average variance extracted estimate revealed that the overall amount of variance in the indicators was accounted for by the latent construct. All average variance explained (AVE) values ranged from 0.730 to 0.830, and were thus acceptable, if following from Fraering and Minor (2006). As indicated in Table 12, all the average value extracted (AVE) are above the shared values (SV) for all the research constructs, for example, the average variance extracted (AVE) of purchase intention (PI), which is 0.80, is greater than the shared variance (SV) of purchase intention (PI) and pricequality inference (PQ), which is 0.752, where this further confirms the existence of discriminant validity.

5.6 VALIDITY TESTS

5.6.1 Convergent Validity

Convergent validity was assessed by checking if individual item loadings for each corresponding research construct were greater than 0.5 as recommended by (Anderson and Gerbing 1988). As indicated in Table 10 (see page 65) the factor loadings ranged from 0.6 to 0.9. All the items used for this study had a loading of more than the recommended 0.5, revealing acceptable individual item convergent validity as more than 50 percent of each item's variance was shared with its respective construct. This finding w supported the convergent validity of all scale items.

5.6.2 Discriminant Validity

5.6.2.1 Correlation Matrix

Another approach used to test for discriminant validity of the research constructs was the evaluation of whether the correlations among latent constructs were less than 1.0. As indicated in Table 10 (see page 65), the inter-correlation values for all paired latent variables were lower than 1.0, therefore positing the presence of discriminant validity. All the latent variables had values less the recommended 0.7 (Nunnally and Bernstein, 1994).

5.6.2.2 Inter-construct correlation matrix

The inter-construct correlation matrix was used to check for discriminant validity of the research constructs. Correlations among latent constructs were evaluated in order to see if

they were lower than 1.0. A value lower than 0.7 for research constructs is recommended to confirm discriminant validity (Nunnally and Bernstein, 1994). The values for the constructs presented below are marginally accepted as they range from 0.6 to 0.7. Below is Table 13 illustrates the relationships among constructs.

Table 13: Correlations between Constructs

Interconstruct Correlation Matrix								
	PBC	ATT	PQ	PI				
Perceived behavioural control PBC	1							
Attitudes towards economic benefits ofpurchasing counterfeit productsATT	.652**	1						
Price – quality inference of purchasing counterfeit products PQ	.658**	.775**	1					
Purchase intention of counterfeit products PI	.721**	.740**	.752**	1				
**. Correlation is significant at the 0.01 level (2	2-tailed).							

5.6.2.3 Average Value Extracted (AVE) and Shared Value (SV)

Discriminant validity was also confirmed by testing if the AVE value was greater than the highest shared variance (S.V.) value (Fornell and Larcker 1981). As indicated in Table 10, all the average variance extracted (AVE) values were above the shared values (SV) for all the research constructs, therefore further confirming the existence of discriminant validity. The average variance extracted (AVE) of (ATT) was 0.833 (see page 65) which is greater that the square of the shared variance of (PQ) and (ATT) which $[(0,775)^2] = 0.601$. This therefore proves the existence of discriminate validity (Nusair & Hua, 2010).

5.7 Confirmatory Factor Analysis (CFA) models

Figure 15 on the following page is an illustration of the confirmatory factor analysis model.





Note:

- PBC= Perceived behavioural control
- ATT= Attitudes towards economic benefits of purchasing counterfeit products
- PQ= Price-quality inference of counterfeit products
- PI= Purchase intention of counterfeit products

The research model was run and had the following results.

Chi-square $(\chi^2/df) = 2.456$, Goodness of Fit Index (GFI) = 0.901, Comparative Fit Index (CFI) = 0.971, Tucker Lewis Index (TLI) = 0.964, Incremental Fit Index (IFI) = 0.971, Relative Fit Index (RFI) = 0.941, Norm Fit Index (NFI) = 0.952, Random Measure of Standard Error Approximation (RMSEA) = 0.062

Table 14 below shows the model fit results.

Table 14: Model Fit Results

Model fit criteria	Chi-square (χ ² /DF)	GFI	CFI	TLI	IFI	RFI	NFI	RMSEA
Indicator								
value	2,46	0.901	0.971	0.964	0.971	0.941	0.952	0.062

5.8. CONCEPTUAL MODEL FIT ASSESSEMENTS

Anderson and Gerbing (1988) proposed a two-step procedure for assessing model fit comprising of confirmatory factor analysis (CFA) and hypotheses testing. Confirmatory factor analysis (CFA) was primarily performed to examine scale accuracy (including reliability, convergent validity, and discriminant validity) of the multiple-item construct measures using AMOS 22. Acceptable model fit was indicated by chi-square value over degree of freedom (χ^2 /df) of value between 1 and 3, the values of Goodness of Fit Index (GFI), Comparative Fit Index (CFI), Incremental Fit Index (IFI) Tucker Lewis Index (TLI), and equal to or greater than 0.90, and the Root Mean Square Error of Approximation (RMSEA) value to be equal to or less than 0.08. Recommended statistics for the final overallmodel assessment revealed acceptable fit of the measurement model to the data. All correlation values were less than 0.8, the measurement model produced a ratio of chi-square value over degree-of-freedom of 2,46, and GFI, CFI, TLI, IFI, RFI, NFI and RMSEA were 0.901, 0.971, 0.964, 0.971, 0.941, 0.952 and 0.062, respectively. Since an acceptable CFA measurement model fit was obtained, the study proceeded to the hypothesis testing stage using structural equation modeling with AMOS 22 software program.

Figure 16: Structural Model



Key:

- PBC= Perceived behavioural control
- PQ = Price-quality inference of counterfeit products
- ATT = Attitudes towards economic benefits of purchasing counterfeit products
- PI = Purchase intention of counterfeit products
- e = measurement error

Table 15: Results of Structural Equation Model Analysis

Proposed relationship hypothesis	Hypothesis	Factor	P Value	Supported/
		Loading		Rejected
Perceived behavioural control (PBC) \longrightarrow	H_1	0.22 ^c	0.01	Supported and
Purchase intention of counterfeits (PI)				significant
Perceived behavioural control (PBC) \longrightarrow	H_2	0.18 ^c	0.01	Supported and
Attitudes towards economic benefits of				significant
purchasing counterfeit products (ATT)				

Price – quality inference of counterfeit products	H ₃	0.82 ^c	0.01	Supported and
$(PQ) \longrightarrow$ Attitudes towards economic benefits				significant
of purchasing counterfeit products (ATT)				
Price – quality inference of counterfeit products	H_4	0.47 ^c	0.01	Supported and
$(PQ) \longrightarrow$ Purchase intention of counterfeits (PI)				significant
Attitudes towards economic benefits of	H5	0.26 ^c	0.01	Supported and
purchasing counterfeit products (ATT) \longrightarrow				significant
Purchase intention of counterfeits (PI)				

Structural model fits: $\chi^2/df = 2,46$; GFI= 0.901; CFI= 0.971; TLI= 0.964; IFI= 0.971; RFI= 0.941; NFI= 0.952;

RMSEA= 0.062

^a significance level p <0.05; ^b significance level <0.01; ^c significance level <0.001

5.10 Model Fit Results

The assessment of the proposed conceptual model proceeded using the same data set. The ratio of chi-square over degree-of-freedom was 2.46. This value is less than the recommended threshold of less than 3.0 and therefore, confirms the model fit (Chinomona, 2011). The GFI, CFI, TLI, IFI, RFI, NFI and RMSEA values were 0.901, 0.971, 0.964, 0.971, 0.941, 0.952 and 0.062. The model fits measures exceeded the recommended acceptable threshold of than 0.8 for GFI, CFI, TLI and for RMSEA the model fit was lower the recommended threshold of 0.08 (Bentler and Bonett, 1980; Bollen, 1989; Baumgartner and Hombur 1996; Byrne 1998; Hu and Bentler, 1999; McDonald and Ho, 2002; Hooper, Coughlan and Mullen, 2008; Chinomona, 2011). These results suggest that the proposed research conceptual model converged satisfactorily and could represent the underlying empirical data structure collected at the University of the Witwaterwatersrand. The model fit was acceptable and the study proceded to test the research hypotheses.

5.11 Discussion of Hypothesis Testing Results

As seen in figure 16 or table 15, all hypotheses coefficients were at least at a significant level of p < 0.01. Hypotheses one, four and five (H1, H4 and H5) posited that purchase intention of

counterfeits had a positive and direct relationship with perceived behavioural control, the price-quality inference of counterfeit products and attitudes towards economic benefits of purchasing counterfeit products. All these three hypotheses were supported, therefore, indicating that the customer's intention to purchase counterfeits has an important and significant impact on the customer's perceptions towards counterfeits.

Hypothesis two (H2) posited that perceived behavioural control is positively and directly related to attitudes towards economic benefits of purchasing counterfeit products. This hypothesis was supported therefore, indicating that the ease or difficulty of performing behaviour has important and significant impact on the perceptions that customers might have towards purchasing counterfeit products. Hypothesis three (H3) posited that attitudes towards economic benefits of purchasing counterfeit products are positively and directly related to price–quality inference of counterfeit products. This hypothesis was supported therefore, indicating that the relationship that exists between the price and quality of a counterfeit products.

5.12 CHAPTER 5 SUMMARY

Chapter 5 is divided into six sections. The chapter provided the empirical results drawn from the study. The first section is the introduction which begins with a brief overview of chapter 4 and then proceeds to explore the data screening process followed by the data analytics procedures. A diagrammatic representation of the statistical analytical procedures is illustrated. The second section explores the descriptive statistics of the current study, which includes a detailed description of the sample, together with a diagrammatic illustration of the sample profile. Thereafter, in the third section, the reliability tests are discussed, including a diagrammatic illustration of the reliability tests. Initial and final analysis accuracy statistics are also illustrated diagrammatically in the third section. The forth section is comprised of the reliability test; the Cronbach's alpha (α), Composite reliability (CR) and the Average variance extracted (AVE). The fifth section is comprised of the validity tests, construct validity, convergent validity and discriminant validity. Also included in the fifth section is a diagrammatic illustration of the inter-construct correlation matrix, a comparison of the average variance extracted (AVE) and the shared variance (SV). Model fit is explored include a diagrammatic representation of model fit criteria and acceptable fit level. The confirmatory factor analysis (CFA) Model is illustrated and discussed. Thereafter, the model fit results are discussed and illustrated diagrammatically. An introduction to structural equation modeling is provided followed by a diagrammatic illustration of the structural model. The structural model is followed by hypothesis testing and a discussion of results of structural equation model analysis. In Chapter 5, the proposed hypothesis for the study was a test requiring conducting structural equation modeling (SEM) using the AMOS 22 software program and then discussion the research findings. The study investigated the direct effects of perceived behavioural control and price quality inference on the attitudes towards economic benefits of purchasing counterfeit products, and the purchase intention of counterfeit products. In a nutshell, all the five posited hypotheses were supported, as expected. Finally the sixth section is made up of the summary of Chapter 5 and a brief discussion of Chapter 6.



Figure 17: Diagrammatic Representation of Chapter 6

CHAPTER 6: FINDINDGS, CONCLUSIONS, IMPILICATIONS AND FUTURE RESEARCH

6.1 INTRODUCTION

This chapter explores the findings of the study, limitations, implications and possible future research. Chapter 6 also provides a comprehensive conclusion of the entire study. The aim of this study was to examine the relationship that perceived behavioural control and price quality inference had with attitudes towards the economic benefits of purchasing counterfeit products and purchase intention of counterfeit products. Structural equation modeling (SEM) was applied to assess the proposed research model and hypothesis. The analytical results implied that all the proposed hypotheses were supported. It was observed that the influence of price-quality inference of counterfeit products (PQ) on attitudes towards economic benefits of purchasing, as it had the highest factor loading, indicated by 0.82 (see Table 15).

6.2 IMPLICATIONS

The current research has both academic and practical implications. Academically, the current study contributes to existing literature on the relationship between perceived behavioural control and purchase intention of counterfeit products. The implication of this relationship is that marketers need to understand the behaviour of counsumers in order to predict whether or not the are likely going to purchase counterfeit products. Still on the academic front it was observed that perceived behavioural control's relationship with attitudes towards the economic benefits of purchasing counterfeit products was positive. This implied that consumer attitudes towards purchasing counterfeits are directly impacted by consumers' perceived ease or difficulty of making the actual purchase therefore marketers of genuine brands have to increase the perceived ease of purchasing genuine products in order to make the purchase of genuine products more attractive for potential consumers.

The relationship between price-quality inference of counterfeit products and attitudes towards the economic benefits of purchasing counterfeit products was also positive. This relationship implied that perceptions of consumers towards counterfeits were a result of the relationship that existing between price and quality of products and thus marketers need to keep the price-quality relationship of genuine products at apropriate levels for most customers so as to attract customers and deter them from purchasing counterfeits. The relationship between price-quality inference of counterfeit products and purchase intention of counterfeit products was also positive. This implied that the ultimate decision for the consumer to purchase a counterfeit product was still determined by the perceived benefit of purchasing a lower priced product regardless of quality. This meant that marketers must ensure that their geniune products are not too expensive as customers will more likely overlook the inferior quality of counterfeits and purchase those counterfeits as they would perceive the relationship between quality and price would to be more balanced in the case of counterfeits.

On the practical front, the current study also has some implications for marketing practitioners in South Africa. According to the findings, price-quality inference of counterfeit products is seen to possess the highest impact on customer attitudes towards the purchase intention of counterfeit products. Price-quality inference in the current study was defined by Zhou et al. (2002) as a product's price cue, that is related positively to the quality level of the product. This would imply that marketers in South Africa should find ways of selling genuine products at prices that are not too high for customers, so as to discourage them from purchasing inferior quality products being offered at lower prices by conterfeit manufacterers.

6.3 FUTURE RESEARCH AND LIMITATIONS

The current study leaves room for further future research in the field of counterfeit products.

There are opportunities for researchers to either explore the same variables of this study in greater depth, or to add new variables that could potentially predict the purchase intention of counterfeit products. The fact that respondents were all registered students implied that they had access to financial resources and education, which may have created a bias in their own perceptions about counterfeit products. Their views can furthermore be generalised to the greater Johannesburg region. The researcher would recommend that similar studies be conducted on a larger scale, that is, with a larger sample size and within larger geographic regions, so as not to be limited to respondents of a certain demographic profile. For instance, the current study only used students from one university, and recommendations for future research would be for researchers to consider larger populations, possibly surveying respondents from more than one university, city or province. This would allow for more diversity in the sample and for more informed results. The study suffered greatly from financial and time constraints, and if this could be addressed in futures studies, the possibility of producing more informed results would be greater.

6.4 OVERALL RESEARCH CONTRIBUTION

The current study makes a contribution to literature on the purchase intention of counterfeit products. Based on the findings of the study, perceived behavioural control has a positive impact on consumers purchase intention of counterfeit products. Based on the definition of perceived behavioural control given by Ajzen (2002), this would imply that human behaviour is responsible for consumers' intent to purchase counterfeit products. According to Ajzen (2002), perceived behavioural control is the independently perceived simplicity, or difficulty of performing a specific behaviour. Another contribution to the literature made here was to confirm that perceived behavioural control has a direct impact on consumers' attitudes towards economic benefits of purchasing counterfeit products, as suggested by the opening hypothesis. A notable contribution to literature made by the current study was to confirm that

price-quality inference of counterfeit products has a direct and positive relationship with consumers' attitudes towards economic benefits of purchasing counterfeit products. This finding is supported by a study conducted by De Matos, Ituassu and Rossi (2007) that also found that the price-quality inference of counterfeit products had a positive relationship with consumers' attitudes towards counterfeit products. Finally, the last contribution to the literature of counterfeit purchases made by the current study is that attitudes towards economic benefits of purchasing counterfeit products proved to have a positive relationship with customers' purchase intention in relation to counterfeit products, as suggested by the hypothesis of the current study. This finding is supported by a study conducted by Yoo and Lee (2009), also in the field of counterfeit purchases. Yoo and Lee (2009) found that attitudes towards towards economic benefits of purchasing counterfeit products had a positive impact on the purchase intention of counterfeit products.

6.6 SUMMARY OF CHAPTER 6

The purpose of the study was to investigate the relationship of the following variables: "perceived behavioural control" and "price-quality inference of counterfeit products" on the following variables: "attitudes towards the economic benefits of purchasing counterfeit products" and "purchase intention of counterfeit products". This was done in an attempt to establish whether the former variables were predictors of the latter variables. Based on the above-mentioned variables, a conceptual research model was developed (see Figure 4). This final chapter provided four concluding sections, presenting a conclusion for the study that made up this thesis, noting the implications of these findings for study. The Chapter then suggested directions of future studies in the field of counterfeit products purchases and closed with an overall conclusion, noting the contributions made to broader enquiry.

LIST OF REFERENCES

Aaker, D.A. (2007), *Marketing research*, 9th edition, New York: John Wiley and Sons.

Ajzen, I., and Fishbein, M. (1980). *Understanding attitudes and predicting social behaviour*. Englewood Cliffs, NJ: Prentice-Hall.

Ajzen, I., and Madden, T. J. (1986). Prediction of goal-directed behaviour: Attitudes, intentions, and perceived behavioral control. *Journal of Experimental Social Psychology*, 22(5), 453-474.

Ajzen, I., and Madden, T. J. (1988). Prediction of goal-directed behaviour: Attitudes, intentions and perceived behavioral control. *Journal of Experimental Social Psychology*, 22, 453-474.

Ajzen, I. (1988). *Attitudes, personality and behaviour*. Milton Keynes, UK: Open University Press.

Ajzen, I. (1991). The theory of planned behaviour. Organisational Behaviour and Human Decision Processes, 50(2), 179-211.

Ajzen (2002) Perceived behavioral control, self-efficacy, locus of control, and the theory of planned behavior, *Journal of Applied Social Psychology*, *32*(4), 665-683.

Ajzen, I. (2005). *Attitudes, personality and behaviour* (2nd ed.). Berkshire: Open University Press and McGraw-Hill.

Aleassa, H., Pearson, J. M., & McClurg, S. (2011). Investigating software piracy in Jordan: An extension of the theory of reasoned action. *Journal of Business Ethics*, *98*(4), 663-676.

Anderson, J. C., and Gerbing, D. W. (1988). Structural equation modeling in practice: A review and recommended two-step approach. *Psychological bulletin*, *103*(3), 411-423.

Ang, S. H., Cheng, P. S., Lim, E. A., and Tambyah, S. K. (2001). Spot the difference: consumer responses towards counterfeits. *Journal of Consumer Marketing*, *18*(3), 219-235.

Armitage, C. J., and Conner, M. (2001). Efficacy of the theory of planned behaviour: A metaanalytic review. *British Journal of Social Psychology*, *40*(4), 471-499.

Bagozzi, R. P., Yi, Y. and Phillips, L. W (1991), Assessing construct validity in organisational Research, *Administrative Science Quarterly*, *36*(3), 421–458.

Bai, B., Law, R., and Wen, I. (2008). The impact of website quality on customer satisfaction and purchase intentions: Evidence from Chinese online visitors. *International Journal of Hospitality Management*, 27(3), 391-402.

Baines, P., Fill, C., and Page, K. (2011). Marketing, 2.,[updated] ed. Oxford University Press. Oxford,UK

Barrett, P. (2007), Structural equation modeling: Adjudging model fit, *Personality and Individual Differences*, 42 (5), 815-24.

Baumgartner, H., & Homburg, C. (1996). Applications of structural equation modeling in marketing and consumer research: A review. *International Journal of Research in Marketing*, *13*(2), 139-161.

Belk, R. W., Devinney, T. and Eckhardt, G. (2005). Consumer ethics across cultures, *Consumption, Markets and Culture*, 8(1), 275-89.

Bentler, P. M., and Bonett, D. G. (1980). Significance tests and goodness of fit in the analysis of covariance structures. *Psychological Bulletin*, *88*, 588-606.

Bentler, P.M. (1990), Comparative fit indexes in structural equation modeling, *Psychological Bulletin*, 107(2), 238-46.

Bian, X. and Moutinho, L. (2011). Counterfeits and branded products: Effects of counterfeit ownership. *Journal of Product and Brand Management*, 20(5), 379-393.

Bollen, K.A. (1989). *Structural Equations with Latent Variables*. New York, USA: John Wiley and Sons.

Boyle, G. J. (1991). Does item homogeneity indicate internal consistency or item redundancy in psychometric scales? *Personality and individual differences*, *12*(3), 291-294.

Budiman, S., & Wijaya, T. (2014). Purchase Intention of Counterfeit Products: The Role of Subjective Norm. *International Journal of Marketing Studies*, *6*(2), 145-152.

Browne, M. W., & Cudeck, R. (1993). Alternative ways of assessing model fit. *Sage Focus Editions*, *154*, 136-136.

Brunetto, Y., and Farr-Wharton, R. (2007). The moderating role of trust in SME owner/managers' decision-making about collaboration. *Journal of Small Business Management*, *45*(3), 362-387.Burton, D. B., Ryan, J. J., Axelrod, B. N., Schellenberger, T., & Richards, H. M. (2003). A confirmatory factor analysis of the WMS-III in a clinical sample with crossvalidation in the standardization sample. *Archives of Clinical Neuropsychology*, *18*(6), 629-641.

Byrne, B. M. (1998). *Structural Equation Modeling with LISREL, PRELIS and SIMPLIS: Basic concepts, applications and programming.* Mahwah, New Jersey: Lawrence Erlbaum Associates.

Byrne, B. M. (2001). *Structural equation modeling with AMOS, basic concepts, applications, and programming*. New Jersey: Lawrence Erlbaum Publishers.

Cademan., A, Henriksson. R, and Nyqvist. V. (2012). The effect of counterfeit products on luxury brands: An empirical investigation from the consumer perspective, Linnaeus University, Sweden, Bachelor Thesis

Calder, B. J., Phillips, L. W., and Tybout, A. M. (1983). Beyond external validity. *Journal of Consumer Research*, 10(1) 112-114.

Campbell, D. T. (1979). Degrees of freedom and the case study. *Qualitative and Quantitative Methods in Evaluation Research*, *1*, 49-67.

Chakraborty, G., Allred, A., Sukhdial, A. S., and Bristol, T. (1997). Use of Negative Cues to Reduce Demand for Counterfeit Products. *Advances in Consumer Research*, *24*, 345-349.

Chaudhry, P. E. and Stumpf, S. A. (2011). Consumer Complicity with Counterfeit Products. *Journal of Consumer Marketing*, 28(2), 139-151.

Chaudary, M. W. T., Ahmed, F., Gill, M. S., and Rizwan, M. (2014). The determinants of purchase intention of consumers towards counterfeit shoes in Pakistan. *Journal of Public Administration and Governance*, *4*(3), 20-38.

Chaudhry, P. E., Zimmerman, A., Peters, J. R., and Cordell, V. V. (2009). Preserving intellectual property rights: Managerial insight into the escalating counterfeit market quandary. *Business Horizons*, *52*(1), 57-66.

Chen, Y. Q., Zhang, Y. B., Liu, J. Y., & Mo, P. (2011). Interrelationships among critical success factors of construction projects based on the structural equation model. *Journal of Management in Engineering*, 28(3), 243-251.

Cheng, S.-I., Fu, H.-H. and Tu, L. T. C. 2011. Examining customer purchase intentions for counterfeit products based on a modified theory of planned behavior. *International Journal of Humanities and Social Science*, *1*(10), 278–284.

Cheung, G. W., and Rensvold, R. B. (2002). Evaluating goodness-of-fit indexes for testing measurement invariance. *Structural Equation Modeling*, *9*(2), 233-255.

Chinomona, R., Lin, J. Y. C., Wang, M. C. H., and Cheng, J. M. S. (2010). Soft power and desirable relationship outcomes: The case of Zimbabwean distribution channels. *Journal of African Business*, *11*(2), 182-200.

Chinomona, R. (2011). Non-mediated channel powers and relationship quality: A case of SMES in Zimbabwe channels of distribution (pp. 1-175) Unpublished PhD. National Central University, Taiwan.

Churchill, G. A. Jr. (1979), A paradigm for developing better measures of marketing constructs, *Journal of Marketing Research*, *16*(1), 64-73.

Churchill, G. A., and Iacobucci, D. (2002). *Marketing Research Methodological Foundations* 8th Edition. USA: South Western, Thompson Learning.

Collins, H. (2010). *Creative research: The theory and practice of research for the creative industries*. Ava Publishing.London, UK

Conner, M., and Sparks, P. (1996). The theory of planned behaviour and health behaviours. In M. Conner and P. Norman (Eds.), *Predicting Health Behaviour* (pp. 121-162). Buckingham, UK: Open University Press.

Conner, M., and Armitage, C. J. (1998). Extending the theory of planned behavior: A review and avenues for further research. *Journal of applied social psychology*, 28(15), 1429-1464.

Cooke, R., and Sheeran, P (2004). Moderation of cognition-intention and cognitionbehaviour relations: A meta-analysis of properties of variables from the theory of planned behaviour. *British Journal of Social Psychology*, *43*(2), 159-186.

Cordell, V. V., Wongtada, N., and Kieschnick, R. L. Jr. (1996). Counterfeit purchase intentions: Role of lawfulness attitudes and product traits as determinants. *Journal of Business Research*, *35*(1), 41-53.

Daniel, J. (2012). Sampling essentials: Practical guidelines for making sampling choices. Sage Publications, London, UK

Deci, E. L. and Ryan, R. M. (2002) *Handbook of Self- Determination Research*. New York: University of Rochester Press.

De Matos, C. A., Ituassu, C. T., and Rossi, C. A. V. (2007). Consumer attitudes toward counterfeits: A review and extension. *Journal of Consumer Marketing*, *24*(1), 36-47.

Dodds, W. B., Monroe, K. B., and Grewal, D. (1991). Effects of price, brand, and store information on buyers' product evaluations. *Journal of Marketing Research*, 28(3) 307-319.

Doss, E. and Robinson, T. (2013). Luxury Perceptions: Luxury Brand vs Counterfeit for Young US Female Consumers. *Journal of Fashion Marketing and Management*, *17*(4), 424-439.

Driscoll, D. L. (2011). Introduction to primary research: Observations, surveys and interviews. In: *Writing Spaces: Readings on Writing*: Parlor Press. Colorado, USA

Dusick, D., (2011). Writing the methodology section. Retrieved from: http://www.bold-ed.com/ch3.htm. Accessed [23/06/2014]

Dunn, S. C., Seaker, R. F. and Waller, M. A. 1994, Latent Variables in Business Logistics
Research: Scale Development and Validation. *Journal of Business Logistics*, *15*(2), 145-172.
Dunn, T. J., Baguley, T., & Brunsden, V. (2014). From alpha to omega: A practical solution
to the pervasive problem of internal consistency estimation. *British Journal of Psychology*, *105*(3), 399-412.

Eagly, A. H., and Chaiken, S. (1993). *The psychology of attitudes*. Harcourt Brace Jovanovich College Publishers. Orlando, Florida, USA

Easterby-Smith, M.T., Thorpe, R. and Lowe, A. (2002), *Management Research: An Introduction*, London: Sage.

Eisend, M., and Schuchert-Güler, P. (2006) Explaining counterfeit purchases: A review and preview. *Academy of Marketing Science Review*, 12:1–22.

European Commission Directorate-General Single Market Centre for Economics and Business Research (2002). *Counting Counterfeits: Defining a method to collect, analyse and compare data on counterfeiting and piracy in the single market* (Final Report). Retrieved from: www.cebr.com. Accessed [10/07/14]

Fagarasanu, M., and Kumar, S. (2002). Measurement instruments and data collection: A consideration of constructs and biases in ergonomics research. *International Journal of Industrial Ergonomics*, *30*(6), 355-369.

Fernandes, C. (2013). Analysis of counterfeit fashion purchase behaviour in the United Arab Emirates. *Journal of Fashion Marketing and Management*, *17*(1), 85-97.

Fishbein, M., and Ajzen, I. (1975). *Belief, attitude, intention and behaviour: An introduction to theory and research*. Reading, Boston, MA:USA: Addison-Wesley

Fornell, C.R. and Larcker, D.F (1981). Evaluating structural equation models with unobservable variables and measurement error. *Journal of Marketing Research*, *18*(1), 39–50.

Fraering, M., and Minor, M. S. (2006). Sense of Community: An Exploratory Study of US Consumers of Financial Services, *International Journal of Bank Marketing* 24(5), 284-306.

Francis, J., Eccles, M. P., Johnston, M., Walker, A. E., Grimshaw, J. M., Foy, R.,

Frankfort-Nachmias, C., and Nachmias, D. (1997). *Research methods in the social sciences*. London, UK: Arnold

Kaner, E. F. S., Smith, L. & Bonetti, D. (2004). Constructing questionnaires based on the theory of planned behaviour: A manual for health services researchers. Newcastle upon Tyne, UK: Centre for Health Services Research, University of Newcastle upon Tyne.

Fukukawa, K. (2002), Developing a framework for ethically questionable behavior in

consumption. Journal of Business Ethics, 47(4), 16-23.

Furnham, A. and Valgeirsson, H. (2007). The effect of life values and materialism on buying counterfeit products. *The Journal of Socio-Economics*, *36*, 677-85.

Galpin, J.S (2011). *Statistical research design and analysis*. [Lecture notes]. Johannesburg, South Africa: University of the Witwatersrand, School of Statistics and Actuarial Science.

Gatignon, H. (2010). Statistical Analysis of Management Data. Boston, MA, USA: Kluwer Academic Publishers

Gentry, J. W., Putrevu, S. and Shultz, C. II (2006), The effects of counterfeiting on consumer research, *Journal of Consumer Behaviour*, 5(3), 245-57.

Grafstrom, A. (2010). Entropy of unequal probability sampling designs. *Statistical Methodology*, *7*, 84-97.

Green, R. T. and Smith, T. (2002), Countering brand counterfeits, *Journal of International Marketing*, *10*(4), 89-99.

Gino, F. M. Norton, and Ariely, D. 2010, The counterfeit self: The deceptive costs of faking it, *Psychological Science*, 21, 712-720

Gistri, G., Romani, S., Pace, S., Gabrielli, V. and Grappi, S. (2009). Consumption Practices of Counterfeit Luxury Goods in the Italian Context. *Journal of Brand Management*, *16*(5/6), 364-374.

Grossman, G. M. and Shapiro , C. (1988). Foreign counterfeiting of status goods. *Quarterly Journal of Economics*, *103*(1), 79 – 100.

Gupta, P. B., Gould, S. J. and Pola, B. (2004), To pirate or not to pirate: A comparative study of the ethical versus the other influences on the consumers' software acquisition-mode decision, *Journal of Business Ethics*, *55*(3), 255-70.

Guo, S., and Feng, G. (2012). Understanding support for internet censorship in China: An elaboration of the theory of reasoned action. *Journal of Chinese Political Science*, *17*(1), 33-52.

Hair, J.F., Anderson, R.E, Tatham, R.L and Black, W.C. (1998) *Multivariate data analysis*. 5th edition. Upper Saddle River, New Jersey: Prentice Hall.

Hair, J.F., Anderson, R.E, Tatham, R.L. and Black, W.C. (2006) *Multivariate data analysis*, 6th edition, London: Prentice-Hall.

Hansen, T., Jensen, J., and Solgaard, H. (2004). Predicting online grocery buying intention: A comparison of the theory of reasoned action and the theory of planned behavior. *International Journal of Information Management*, *24*(6), 539-550.

He, Y., Gai, Y., Wu, X., and Wan, H. (2012). Quantitatively analyze composition principle of Ma Huang Tang By structural equation modelling. *Journal of Ethno pharmacology*, *143*, 851-858.

Heffes, Ellen M. (2008). "Fending Off Pirates," Financial Executive, 24 (March), 40-42.

Henle, C. A., Reeve, C. L., and Pitts, V. E. (2010). Stealing time at work: Attitudes, social pressure, and perceived control as predictors of time theft. *Journal of Business Ethics*, *94*(1), 53-67.

Hieke, S. (2010). Effects of Counterfeits on the Image of Luxury Brands: An Empirical Study from The Customer Perspective. *Brand Management*, *18* (2). *159-173*.

Hilton, B., Choi, C.J. and Chen, S. (2004), "The ethics of counterfeiting in the fashion industry: quality, credence and profit issues", Journal of Business Ethics, Vol. 55 No. 4, pp. 345-54.

Hirschi, T. (2004). Self-control and crime. In R. F. Baumeister and K. D. Vohs (Eds.), Handbook of self-regulation: Research, theory, and applications (pp. 537-552). New York, NY: Guilford.

Hirschman, E. C. (1986). Humanistic inquiry in marketing research: philosophy, method, and criteria. *Journal of marketing Research*, 23(3) 237-249.

Hoecht, A., & Trott, P. (2014). How should firms deal with counterfeiting? A review of the success conditions of anti-counterfeiting strategies. *International Journal of Emerging Markets*, *9*(1), 98-119.

Hooper, D., Coughlan, J., and Mullen, M. (2008). Structural equation modeling: Guidelines for determining model fit. *Electronic Journal of Business Research Methods*, 6(1), 53-60.

Houran, J., & Lange, R. (2013). Applying The Theory of Reasoned Action To A Computerized Test of Intuition, Part II: Decision-Making In A Hidden Test of PSI. *Journal of the Society for Psychical Research*, 77(913), 236-251

Howie, M. (2010, November 10). *Fake goods are fine, says EU study*. Telegraph. Retrieved from: <u>http://www.telegraph.co.uk/finance/newsbysector/retailandconsumer/7969335/Fake-goods-are-fine-says-EU-study.html</u>. Accessed [23/06/2014]

Hu, L. T., & Bentler, P. M. (1999). Cutoff criteria for fit indexes in covariance structure analysis: Conventional criteria versus new alternatives. *Structural equation modeling: a multidisciplinary journal*, *6*(1), 1-55.

Huang, J. H., Lee, B. C. Y. and Ho, S. H. (2004), Consumer attitude toward gray market goods, *International Marketing Review*, *21*(6), 598-614.

Hsu, J. L., and Shine, C. W. (2008). Consumers' willingness to pay for non-pirated software. *Journal of Business Ethics*, *81*(4), 715-732.

Hung, C.L. (2003), "The business of product counterfeiting in China and the post-WTO membership environment", Asia Pacific Business Review, Vol. 10(1) 58-77.

Husic, M., and Cicic, M. (2009). Luxury consumption factors. *Journal of Fashion Marketing and Management*, *13*(2), 231-245.

ImportantFactsAboutWits.Retrievedfrom:http://www.wits.ac.za/prospective/undergraduate/11637/what_parents_need_to_know.html.Accessed [2014/ 03/ 2014]

International Chamber of Commerce (2008). *Luxury goods: Facts and figures*. Retrieved from: http://wwwiccwboorg/bascap/id6708/indexhtml. Accessed [2014/ 03/ 2014]

Inkon, K. (2013). A Study on Luxuries Possession Desires and Purchase Intention: A Comparative Study between Luxuries and Imitations. *Academy of Entrepreneurship Journal*, *19*(3), 63-77.

Jöreskog, K. G., & Sörbom, D. (1993). *LISREL 8: Structural equation modeling with the SIMPLIS command language*. Scientific Software International.

Jöreskog, K. and Sörbom, D. (1996), *Lisrel 8: User's reference guide*. Chicago, IL: Scientific Software International Inc.

Kaplan, D. (2000). *Structural equation modeling: Foundations and extensions* (Advanced Quantitative Techniques in the Social Sciences Series, 10) (1st ed). Thousand Oaks: Sage Publications.

Kelley, T.L. (1942). The reliability coefficient. Psychometrika, 7(2), 75-83.

Kim, S., Jeong, S. H., and Hwang, Y. (2012). Predictors of pro-environmental behaviors of American and Korean students: The application of the theory of reasoned action and protection motivation theory. *Science Communication*,.

Klein, E.A., and Meyskens, F.L. (2001). Potential target populations and clinical models for testing chemopreventive agents. *Urology*, 57(4), 171-173.

Kraft, P., Rise, J., Sutton, S., and Røysamb, E. (2005). Perceived difficulty in the theory of planned behaviour: Perceived behavioural control or affective attitude? *British Journal of Social Psychology*, *44*(3), 479-496.

Krechevsky, C. (2000), The multinational approach to anti-counterfeiting, *Journal of Proprietary Rights*, *12*(9), 2-13.

Kwong, K.K., Yau, H.M., Lee, S.Y., Sin, Y.M. and Tse, C.B. 2003. The effects of Attitudinal and demographic factors on intention to buy pirated CDs: The case of Chinese consumers. *Journal of Business Ethics*, *47*, 223-235.

Lambkin, M., and Tyndall, Y. (2009). Brand Counterfeiting: A Marketing Problem That Won't Go Away. *Irish Marketing Review*, 20(1), 35-46.

Leisen, B., and Nill, A. (2001). Combatting product counterfeiting: An investigation into the likely effectiveness of a demand-oriented approach. *American Marketing Association Conference Proceedings*, 12, (1), 271-277.

Liao, C., Lin, H. N., and Liu, Y. P. (2010). Predicting the use of pirated software: A contingency model integrating perceived risk with the theory of planned behavior. *Journal of Business Ethics*, *91*(2), 237-252.

Liao, C. H., and Hsieh, I. Y. (2013). Determinants of consumer's willingness to purchase gray-market smartphones. *Journal of business ethics*, *114*(3), 409-424.

Lichtenstein, D. R., Ridgway, N. M., and Netemeyer, R. G. (1993). Price perceptions and consumer shopping behaviour: A field study. *Journal of Marketing Research*, 30(2) 234-245.

Logsdon, J.M., Thompson, J.K. and Reid, A. (1994), Software piracy: Is it related to level of moral judgment?, *Journal of Business Ethics*, *13*(11), 849-57.

Lynch, J. G. (1983). The role of external validity in theoretical research. *Journal of Consumer Research*, 10(1) 109-111.

Lysonski, S. and Durvasula, S. (2008), Digital piracy of MP3: Consumer and ethical predispositions", Journal of Consumer Marketing, 25(3), 167-78.

Malhotra, N. K. (1993). *Marketing research: An applied orientation*. Englewood Cliffs, New Jersey: Prentice-Hall, Inc.

Malhotra, N.K. (1996), Marketing research: An applied orientation, 2nd edition. New Jersey: Simon and Schuster.

Malhotra, N., and Birks, D. (2007). *Marketing Research: an applied approach:* 3rd European edition. Pearson Education: India

Mangan, J., Lalwani, C., and Gardner, B. (2004). Combining quantitative and qualitative methodologies in logistics research. *International Journal of Physical Distribution and Logistics Management*, *34*(7), 565-578.

Mathieson, K., Peacock, E., and Chin, W. W. (2001). Extending the technology acceptance model: The influence of perceived user resources. *ACM SigMIS Database*, *32*(3), 86-112. McDonald, R. P., and Ho, M. H. R. (2002). Principles and practice in reporting statistical

equation analyses. *Psychological Methods*, 7(1), 64-82.

Michaelidou, N., and Christodoulides, G. (2011). Antecedents of attitude and intention towards counterfeit symbolic and experiential products. *Journal of Marketing Management*, 27(9-10), 976-991.

Minagawa, T., Trott, P., and Hoecht, A. (2007). Counterfeit, imitation, reverse engineering and learning: Reflections from Chinese manufacturing firms. *Rand Management*, *37*(5), 455-467.

Mishra, D., Akman, I., and Mishra, A. (2014). Theory of reasoned action application for green information technology acceptance. *Computers in Human Behavior*, *36*, 29-40.

Morris, R. G., Gerber, J., and Menard, S. (2011). Social bonds, self-control, and adult criminality: A nationally representative assessment of Hirschi's revised self-control theory. *Criminal Justice and Behavior*, *38*(6), 584-599.

Myles, G., and Nusser, S. (2006). Content protection for games. *IBM Systems Journal*, 45(1), 119-143.

Naim, M. (2005). Illicit: How Smugglers. *Traffickers, and Copycats are Hijacking the Global Economy*. London: William Heineman

Nevitt, J., and Hancock, G. R. (2000). Improving the root mean square error of approximation for nonnormal conditions in structural equation modeling. *The Journal of Experimental Education*, 68(3), 251-268.

Nia, A., and Zaichkowsky, J. L. (2000). Do counterfeits devalue the ownership of luxury brands? *Journal of Product and Brand Management*, 9(7), 485-497.

Norum, P. and Cuno, A. (2011). Analysis of the demand for counterfeit goods, Journal of

Fashion Marketing and Management, 15(1), 27-40.

Nunnally, J. C. and Bernstein, I. (1994). Psychometric theory, 3rd edition. New

York: McGraw-Hill.

Nunnally, J.C. (1978), Psychometric theory, 2nd edition, New York: McGraw-Hill.

Nusair, K., and Hua, N. (2010). Comparative assessment of structural equation modeling and multiple regression research methodologies: E-commerce context. *Tourism Management*, *31*(3), 314-324.

OECD (2007) *Economic impact of counterfeiting and piracy: Executive summary*. Retrieved from: http//: www.oecd.org/sti/38707619. [Accessed 24/10/2014]

Olson, J. M., and Zanna, M. P. (1993). Attitudes and attitude change. *Annual Review* of *Psychology*, 44(1), 117-154.

Phau, I., Prendergast, G. and Chuen, L. H. (2001). Profiling brand-piracy-prone consumers: An exploratory study in Hong Kong's clothing industry. *Journal of Fashion Marketing and Management*, 5(1), 45-55.

Penz, E., and Stottinger, B. (2005). Forget the "Real" Thing-Take the Copy! An Explanatory Model for the Volitional Purchase of Counterfeit Products. Advances in Consumer Research, *32*(1), 68-575

Penz, E., Schlegelmilch, B. B., and Stöttinger, B. (2008). Voluntary purchase of counterfeit products: Empirical evidence from four countries. *Journal of International Consumer Marketing*, *21*(1), 67-84.
Perez, M. E., Castano, R. and Quintanilla, C. (2010). Constructing Identity through the Consumption of Counterfeit Luxury Goods. *Qualitative Market Research: An International Journal*, *13*(3), 219-235.

Peter, J.P. (1981). Construct validity: A review of basic issues and marketing practice. *Journal of Marketing Research*, 18(2), 113-45.

Petty, N. J., Thomson, O. P., & Stew, G. (2012). Ready for a paradigm shift? Part 2: Introducing qualitative research methodologies and methods. *Manual therapy*, *17*(5), 378-384.

Phau, I., Sequeira, M., and Dix, S. (2009). To buy or not to buy a "counterfeit" Ralph Lauren polo shirt: The role of lawfulness and legality toward purchasing counterfeits. *Asia-Pacific Journal of Business Administration*, *1*(1), 68-80.

Phau, I., Teah, M., and Lee, A. (2009). Targeting buyers of counterfeits of luxury brands: A study on attitudes of Singaporean consumers. *Journal of Targeting, Measurement and Analysis for Marketing*, *17*(1), 3-15.

Poddar, A., Foreman, J., Banerjee, S. S., and Ellen, P. S. (2012). Exploring the Robin Hood effect: Moral profiteering motives for purchasing counterfeit products. *Journal of Business Research*, *65*(10), 1500-1506.

Polit, D.F., and Hungler, B.P. (1995). *Nursing research principles and methods*. 3rd edition. (13-20). Melbourne: Churchill Livingstone.

Prapavessis, H., Gaston, A., & DeJesus, S. (2015). The theory of planned behavior as a model for understanding sedentary behavior. *Psychology of Sport and Exercise*, *19*, 23-32.

Quin, Y (2014) Brand Management and Strategies Against Counterfeits Journal of Economics and Management Strategy, 23(2), 317–343.

Qureshi, S. M., & Kang, C. (2015). Analysing the organizational factors of project complexity using structural equation modelling. *International Journal of Project Management*, 33(1), 165-176.

Riquelme, H. E., Mahdi Sayed Abbas, E., & Rios, R. E. (2012). Intention to purchase fake products in an Islamic country. *Education, Business and Society: Contemporary Middle Eastern Issues*, 5(1), 6-22.

Roche, M., Duffield, C., & White, E. (2011). Factors in the practice environment of nurses working in inpatient mental health: A partial least squares path modeling approach. *International journal of nursing studies*, *48*(12), 1475-1486.

Samgir, H. B., Sharma, P. U., and Sharma, K. (2013). A Study of Customer Response to Brand Imitation, in Pune city. *IJSEAT*, *1*(2), 031-035.

Kneale, J., & Santy, J. (1999). Critiquing qualitative research. *Journal of Orthopaedic Nursing*, *3*(1), 24-32.

Sarstedt, M., Ringle, C. M., Smith, D., Reams, R., and Hair Jr, J. F. (2014). Partial least squares structural equation modeling (PLS-SEM): A useful tool for family business researchers. *Journal of Family Business Strategy*, *5*(1), 105-115.

Schiffman, L. G., Kanuk, L. L., and Wisenblit, J. (2010). *Consumer behavior*. 10th edition. Upper Saddle River, NJ: Pearson Education, Inc.

Schumacker, R. E., & Lomax, R. G. (2004). *A beginner's guide to structural equation modeling*. New York: Psychology Press.

Sheppard, B. H., Hartwick, J., and Warshaw, P. R. (1988). The theory of reasoned action: A meta-analysis of past research with recommendations for modifications and future research. *Journal of Consumer Research*, 15 (3) 325-343.

Shultz, C. J., & Saporito, B. (1996). Protecting intellectual property: Strategies and recommendations to deter counterfeiting and brand piracy in global markets. *The Columbia Journal of World Business*, *31*(1), 18-28.

Staake, T., Thiesse, F., and Fleisch, E (2009), The emergence of counterfeit trade: A literature review. *European Journal of Marketing*, *43*(3/4), 320-349.

Steiger, J. H., and Lind, J. C. (1980). Statistically based tests for the number of common factors. *Annual meeting of the Psychometric Society, Iowa City, IA*, 758, 424-453.

Steiger, J. H. (1998). A note on multiple sample extensions of the RMSEA fit index Structural Equation Modeling: A Multidisciplinary Journal 5, (4), 411-419

Stravinskiene, J., Dovaliene, A., and Ambrazeviciute, R. (2014). Factors influencing intent to buy counterfeits of Luxury Goods. *Economics and Management*, *18*(4), 761-768.

Tabachnick, B.G., and Fidell, L.S. (2007). Using Multivariate Statistics. 5th edition. New York: Allyn and Bacon.

Tan, B. (2002). Understanding consumer ethical decision making with respect to purchase of pirated software. *Journal of Consumer Marketing*, *19*(2), 96-111.

Tavakol, M., and Dennick, R. (2011). Making sense of Cronbach's alpha. *International Journal of Medical Education*, *2*, 53-55.Triandewil, E. and Tjiptono, F. (2013). Consumer Intention to Buy Original Brands versus Counterfeits, International Journal of Marketing Studies, 5(2), 2013.

Trochim, W. (2000). The Research Methods Knowledge Base, 2nd edition. Atomic Dog: OH, Cincinnati Publishing

Tucker, L. R., and Lewis, C. (1973). A reliability coefficient for maximum likelihood factor analysis. *Psychometrika*, *38*(1), 1-10.

Tucker, L. R., and MacCallum, R. C. (1997). *Exploratory factor analysis*. Unpublished PhD thesis, Ohio State University, Columbus, Ohio.

Turunen, L. L. M., and Laaksonen, P. (2011). Diffusing the boundaries between luxury and counterfeits. *Journal of Product and Brand Management*, 20(6), 468-474.

Ullah, (2010). Data Mining Algorithms and Medical Sciences, *International Journal of Computer Science and Information Technology*, 2(6), 127-136

U.S. Customs and Border Protection. (2006). L.A. Strategic trade center. Retrieved from: https://www.cbp.gov. 3/09/2014.

Wang, Y., and Song, Y. (2013). Counterfeiting: Friend or foe of luxury brands? An examination of Chinese consumers' attitudes toward counterfeit luxury brands. *Journal of Global Marketing*, 26(4), 173-187.

Washington, S. P., Karlaftis, M. G., and Mannering, F. L. (2003). *Statistical and Econometric Methods for Transportation Data Analysis*. Chapman & Hall/CRC, Boca Raton, 2nd edition Fla, USA Webb, C. (1989). Action research: Philosophy, methods and personal experiences. *Journal of Advanced Nursing*, *14*(5), 403-410.

Wilckens, T. (2010). Securing reliability and validity in biomedical research: An essential task. *Drug Discovery Today*, *15*(23-24), 991-996.

Wilcox, K., Kim, H. M., & Sen, S. (2009). Why do consumers buy counterfeit luxury brands? *Journal of Marketing Research*, *46*(2), 247-259.

Wilke, R. and Zaichkowsky, J. L. (1999). Brand imitation and its effects on innovation,

competition, and brand equity, Business Horizons, 42(6), 9-18.

Wothke, W. (2010). Introduction to structural equation modeling course notes. *American Institute of Research. SAS Institute Inc. Book Code E*, 1565.

World Customs Organisation (2004). *Counterfeiting Congress Calls for Public-private Cooperation*. Retrieved from: http://www.wcoomd.org/ie/En/Press/press.html.

Veloutsou, C., and Bian, X. (2008). A cross-national examination of consumer perceived risk in the context of non-deceptive counterfeit brands. *Journal of Consumer Behaviour*, 7(1), 3-20.

Yang, H. L., and Lai, C. Y. (2010). Motivations of Wikipedia content contributors. *Computers in Human Behavior*, *26*(6), 1377-1383.

Yoo, B., and Lee, S. H. (2009). Buy genuine luxury fashion products or counterfeits. *Advances in Consumer Research*, *36*(1), 280-228.

Zhan, L., & He, Y. (2012). Understanding luxury consumption in China: Consumer perceptions of best-known brands. *Journal of Business Research*, 65(10), 1452-1460.

APPENDIX A



Participant Information Sheet

Good day,

My name is Tinashe Chuchu, student number 731094, and I am currently completing my Masters in Commerce (Marketing) at the University of the Witwatersrand, Johannesburg.

My current research is entitled "Predictors of Customer Purchase Intention of Counterfeit Products in Johannesburg". Through my research, I aim to explore customer behaviour, perceptions and attitudes that customers in the city of Johannesburg have towards the purchase of counterfeit products. The findings of this study will contribute to the literature of counterfeit products purchase intention.

I am inviting you to be a participant in my current research study. Your selection into this research was based on the fact that you are a potential customer of products in the city of Johannesburg. By being a participant in this research study I would request you to complete the questionnaire attached to this information sheet. With your permission, I ask that you complete this questionnaire and kindly return it to the researcher. This should not take more than 8 minutes of your time.

Your participation in this research is voluntary and I can guarantee that your personal details will remain anonymous throughout this research study as well as in the final research dissertation. You as the participant may refuse to answer any questions which you feel uncomfortable with and may also feel free to withdraw from this study at any time. By being a participant in this research you will not receive payment of any form and the information you disclose will be used in the research report.

This research will be written into a Masters Dissertation and will be available through the University's website. Should you require a summary of the research, the researcher can make this available to you.

Should you have any further questions or queries you are welcome to contact the researcher or the Supervisor, Professor Richard Chinomona at any time at the contact details provided below.

Researcher	<u>Supervisor</u>
Mr Tinashe Chuchu	Professor Richard Chinomona
731094@students.wits.ac.za	richard.chinomona@wits.ac.za
Tel: 0746920263	

Consent Form for completing the research questionnaire

I, _____, acknowledge that I understand the research and that the research has been fully explained to me. I also understand that the information which I give to the researcher will be used in the research report.

I further acknowledge that the researcher has promised me the following:

- That my participation in this research is voluntary
- That my personal details will remain anonymous throughout the research study as well as in the research dissertation
- That I can refuse to answer any questions which I feel uncomfortable with

I hereby consent to completing the questionnaire provided for the research study, "Predictors of Customer Purchase Intention of Counterfeit Products in Johannesburg."

I agree / I do not agree to the completing the questionnaire.

Participant

Researcher

Date Signed

QUESTIONNAIRE

Predictors of Customer Purchase Intention of Counterfeit Products in Johannesburg

Please answer the following questions by marking the appropriate answer(s) with an X. This questionnaire is strictly for research purpose only.

SECTION A: GENERAL INFORMATION

This section is asking about your background information. Please indicate your answer by ticking (X) on the appropriate box.

A1 Please indicate your gender

Male Female

A2 Please indicate your marital status

Married Single

A3 Please indicate your age category

A4 Please indicate your highest academic level

High School	
Diploma	
Degree	
Post graduate degree	
Other (Specify)	

A5 Please indicate your occupation

Student	
Employed	
Self-employed	
Unemployed	
Other (Specify)	

A6 Please indicate the frequency of your luxury product purchasing

Monthly	
Seasonally	
Annually	
Other (Specify)	

A7 Please indicate the amount of money spent on luxury product purchases

Less than R250	
Between R250- R500	
Between R501- R1000	
More than R1000	

Below are statements about Perceived behavioural control, Price-quality Inference of purchasing counterfeit products, Attitudes towards economic benefits of purchasing counterfeit products and Purchase intention measured on a 1 to 7 point likert scale.

SECTION B PERCEIVED BEHAVIOURAL CONTROL

Please indicate to what extent you agree/disagree with each statement as the statement relates to your perceived ease or difficulty of performing the behaviour (Perceived Behavioural Control).

	1 = Disagree completely, + = 1) and $7 = Agree completely$							
1	For me to purchase a counterfeit product over the next two weeks would be difficult.	1	2	3	4	5	6	7
2	If I wanted to, I would not have problems in succeeding to purchase a counterfeit product over the next two weeks.	1	2	3	4	5	6	7
3	I have full control over purchasing counterfeit products over the next two weeks	1	2	3	4	5	6	7
4	It is completely up to me whether or not I purchase a counterfeit product over the next two weeks.	1	2	3	4	5	6	7
5	It is first and foremost up to me whether or not I purchase a counterfeit product over the next two weeks.	1	2	3	4	5	6	7
	1-Very difficult 4- Neutral and 7-Very easy							

1 = **Disagree completely**, 4 = **Neutral** and 7 = **Agree completely**

	1= very uniferret, 4- Neutral and 7- very easy							
6	How easy or difficult would it be for you to purchase a counterfeit	1	2	3	4	5	6	7
	product over the next two weeks?							

1= Completely unconfident, 4= Neutral and 7= Completely Confident

7	How confident are you that you could purchase a counterfeit	1	2	3	4	5	6	7
	product over the next two weeks?							

	1=Very unlikely, 4= Neutral and 7= Very likely							
8	If you actually tried, how likely is it that you would succeed in purchasing a counterfeit product over the next two weeks?	1	2	3	4	5	6	7
							· · · · · ·	

	1= No control at all, 4= Neutral and 7= Complete control							
9	How much control do you feel over purchasing counterfeit products	1	2	3	4	5	6	7
	over the next two weeks?							

SECTION C: ATTITUDES TOWARDS ECONOMIC BENEFITS OF PURCHASING COUNTERFITS

1= Strongly disagree, 4=Neutral and 7= Strongly agree

10	I buy counterfeit products if I think genuine designer products are too	1	2	3	4	5	6	7
	expensive.							
11	I buy counterfeit products if I cannot afford to buy designer products.	1	2	3	4	5	6	7
12	I buy counterfeit products without hesitation if I have a chance to buy	1	2	3	4	5	6	7
	the counterfeits.							
13	I buy counterfeit products, instead of the designer products, if I prefer	1	2	3	4	5	6	7
	specific brands.							
14	I boast about counterfeit products as if they are the genuine brand	1	2	3	4	5	6	7
	products.							
15	I usually purchase counterfeits when it is difficult to distinguish between	1	2	3	4	5	6	7
	the counterfeits and the genuine products.							

SECTION D: PRICE – QUALITY INFERENCE (relationship between the price and quality of products)

16	Generally speaking, the higher the price of a product, the higher the	1	2	3	4	5	6	7
	quality.							
17	The price of a product is a good indicator of its quality.	1	2	3	4	5	6	7
18	You always have to pay a bit more for the best.	1	2	3	4	5	6	7

1= Strongly disagree, 4=Neutral and 7= Strongly agree

SECTION E: PURCHASE INTENTION

1=Very low, 4= Neutral and 7=Very high

1	1	Z	3	4	3	0	/
eit product is	1	2	3	4	5	6	7
ider buying a counterfeit product is	1	2	3	4	5	6	7
of me purchasing a counterfeit product	1	2	3	4	5	6	7
I would consider buying a model at the	1	2	3	4	5	6	7
	Teit product is sider buying a counterfeit product is of me purchasing a counterfeit product I would consider buying a model at the	Teit product is1sider buying a counterfeit product is1of me purchasing a counterfeit product1I would consider buying a model at the1	Teit product is12sider buying a counterfeit product is12of me purchasing a counterfeit product12I would consider buying a model at the12	Teit product is123sider buying a counterfeit product is123of me purchasing a counterfeit product123I would consider buying a model at the123	Teit product is1234Sider buying a counterfeit product is1234of me purchasing a counterfeit product1234I would consider buying a model at the1234	Teit product is12345sider buying a counterfeit product is12345of me purchasing a counterfeit product12345I would consider buying a model at the12345	Teit product is123456sider buying a counterfeit product is123456of me purchasing a counterfeit product123456I would consider buying a model at the123456

If you have any comments please write them below

Thank you for completing the questionnaire. All information will be treated as confidential and not disclosed without your discretion.

APPENDIX B: ETHICS CLEARANCE CERTIFICATE



HUMAN RESEARCH ETHICS COMMITTEE (NON-MEDICAL) R14/49 Chuchu

CLEARANCE CERTIFICATE	PROTOCOL NUMBER H14/07/06				
PROJECT TITLE	Predictors of customer purchase intention of counterfeit products Johannesburg				
INVESTIGATOR(S)	Mr T Chuchu				
SCHOOL/DEPARTMENT	Economic & Business Sciences				
DATE CONSIDERED	18 July 2014				
DECISION OF THE COMMITTEE	Approved Unconditionally				
EXPIRY DATE	29/07/2016				
DATE 30/07/2014	CHAIRPERSON E. M. Tale PP (Professor T Milani)				
cc: Supervisor : Prof R Chinomona					
DECLARATION OF INVESTIGATOR(S)					
To be completed in duplicate and ONE COPY returned to the Secretary at Room 10000, 10th Floor, Senate House, University.					

I/We fully understand the conditions under which I am/we are authorized to carry out the abovementioned research and I/we guarantee to ensure compliance with these conditions. Should any departure to be contemplated from the research procedure as approved I/we undertake to resubmit the protocol to the Committee. <u>Lagree to completion of a yearly progress report.</u>

8 Signature

<u> 011 0812014</u> Date

PLEASE QUOTE THE PROTOCOL NUMBER ON ALL ENQUIRIES